



भारत सरकार

Government of India

विद्युत मंत्रालय

Ministry of Power

केन्द्रीय विद्युत प्राधिकरण

Central Electricity Authority

विद्युत प्रणाली योजना एवं मूल्यांकन-। प्रभाग

Power System Planning & Appraisal-I Division

सं.: के.वि.प्रा./100/3/SBD/2017/PSP&A-I/ 626-724

दिनांक /Date: 14.07.2017

सेवा में /To

- As per list / सुची के अनुसार -

Subject : Comments of stakeholders on Implementation Agreement and Authorization Agreement related to standard bidding documents for procurement of Inter-state transmission service

विषयः अंतर-राज्यीय पारेषण सेवाओ की खरीद के लिए मानक बोली दस्तावेजों से संबंधित कार्यान्वयन समझौता और प्राधिकरण करार पर हितधारकों की टिप्पणियां

महोदय / Sir,

MoP vide its letter dated 11.07.2017 has forwarded the following documents for consultations with all the stakeholders :

- (i) Draft Implementation Agreement (IA) between Transmission Project Developer selected through TBCB and developer of upstream/downstream transmission system
- (ii) Draft Authorization agreement between DICs and counterparty/ nodal agency so as to enable counterparty/ nodal agency to act on behalf of all DICs.

In this regard, you are kindly requested to furnish your valuable comments/ suggestions / modifications on the above documents within 15 days. These documents are available on CEA website on the following link: http://www.cea.nic.in/pspandai.html

Comments/ suggestions / modifications could be sent at the following email addresses: nitindeswal@nic.in and pspandpal.cea@gmail.com

भवदीय / Yours faithfully,

(अवधेश कुमार यादव / Awdhesh Kumar Yadav)

Director, PSP&A-I

Copy to:

1. Under Secretary, Ministry of Power, Shram Shakti Bhawan, New Delhi - 110001.

List of Addresses:

All Power Distribution Utilities

Chief Engineer Haryana Power Purchase Centre 2nd Floor, Shakti Bhawan Sector- 06, Panchkula - 134109	2. Chief Engineer (PP&R) Punjab State Power Corporation Ltd. The Mall, Patiala - 147001	3. Director (Power Trading) Ajmer Vidyut Vitran Nigam Ltd Old Power House, Hathi Bhatta Jaipur Road, Ajmer - 305001
 Managing Director Jodhpur Vidyut Vitran Nigam Ltd. New Power House Industrial Area Jodhpur - 342003 	 Managing Director Jaipur Vidyut Vitran Nigam Ltd. Vidyut Bhawan, Janpath, Jyoti Nagar Jyoti Marg, Jaipur - 302005 	6. Chief Engineer Power Department Govt. of Jammu & Kashmir Civil Sectt., Jammu - 180001
 Asst. Vice President (PMG) BSES Yamuna Power Ltd. 2nd Floor, B Block, Shakti Kiran Building, New Delhi - 110092 	8. Head Power Management Group BSES Rajdhani Power Ltd. BSES Bhawan, 2nd Floor, B-Block Nehru Place, New Delhi' - 110019	 Chief Power- Mgt Tata Power Delhi Distribution Ltd. 33 kV S/Stn. Building, Hudson Lane Kingsway Camp, New Delhi -110019
 Director (Power) New Delhi Municipal Corporation Palika Kendra, Sansad Marg New Delhi - 110001 	11. Chief Engineer (Commercial) UTC - Chandigarh Div-11, Opposite Transport Nagar, Indl. Area Ph-I, Chandigarh - 160002	12. Chief Engineer (PPA) U.P. Power Corporation Ltd. Shakti Bhawan, 14, Ashok Marg Lucknow (U.P.) – 226001
13. SE (Commercial) Uttrakhand Power Corporation Ltd. Urja Bhawan, Kanwali Road Dehradun - 248001	14. Chief Engineer (SOP) Himachal Pradesh State Electricity Board, Vidyut Bhawan Shimla (H.P.) - 171004	15. Secretary (Power) Goa Electricity Department Assembly Complex, Alto Porvorim Penha De Franca, Goa – 403521
16. Chairman Chhattisgarh State Power Distribution Co. Ltd. (CSPDCL) Vidyut Seva Bhavan, Dangania Raipur (Chhattisgarh) - 492013	17. Deputy Engineer DNH Power Distribution Corporation Limited Vidyut Bhavan, Near Secretariat, Amli, Silvassa (DNH) - 396230	18. Executive Engineer Electricity Department Administration of Daman & Diu Secretariat, Fort Area Moti Daman (D&D) - 396220
19. Chairman Madhya Pradesh Power Management Company Ltd. (MPPMCL) Block No-11, Ground floor Shakti Bhawan, Vidhyut Nagar Rampur, Jabalpur (M.P) - 482008	20. Chairman Gujarat Urja Vikas Nigam Limited Energy & Petrochemicals Dept. Block No- 5, 5th Floor New Sachivalay Gandhinagar - 382010	21. Chairman & Managing Director Maharashtra State Electricity Distribution Company Ltd. (MSEDCL) Hongkong Bank Building M.G. Road Fort, Mumbai - 400001
22. Executive Director Maithon Power Limited MA-5, Gogna, PO - Maithon Dam Distt. Dhanbad (Jharkhand) - 828207	23. Director (Commercial) Grid Corporation of Orissa Limited Bhoi Nagar, Janpath Bhubaneshwar (Odisha) - 751022	24. Director (Project) North Bihar Power Distribution Company Limited 3rd Floor, Vidyut Bhawan Bailey Road, Patna (Bihar) - 8000
 Director (Project) South Bihar Power Distribution Company Limited 2nd Floor, Vidyut Bhawan, Bailey Road, Patna - 800001 	26. Chief Engineer (C&R) Jharkhand Bijli Vitran Nigam Ltd. Engineer's Building, Dhurwa Ranchi (Jharkhand) - 834004,	27. Chairman Damodar Valley Corporation (DVC) DVC Headquarters, DVC Towers. VIP Road, Kolkata (W.B.) - 700054
28. PCE cum Secretary Energy and Power Department Govt. of Sikkim Sonam Tshering Marg, Vishal Gaon, Gangtok (Sikkim)- 737102	29. Chief Engineer (PTP) West Bengal State Electricity Distribution Company Limited Vidyut Bhawan, Block- DJ Kolkata (W.B.) - 700091	30. Chairman & Managing Director Northern Power Distribution Co. of Telangana Ltd. (TSNPDCL) H No.2-5-3 1/2, Vidyut Bhawan Nakkal Gutta, Hanamkonda Warrangal - 506001
S1. Chairman & Managing Director Southern Power Distribution Co. of Telangana Limited (TSSPDCL) 6-1-50, Mint Compound Hyderabad - 500063	32. Chairman & Managing Director Southern Power Distribution Co. of Andhra Pradesh Limited (APSPDCL) D.No 19-13-65/A, Srinivasapuram Tirupati (A.P.) - 517503	33. Chairman & Managing Director Eastern Power Distribution Co. of Andhra Pradesh Ltd. (APEPDCL) P&T Colony, Seethammadhara Visakhapatnam - 530013
34. Managing Director Bangalore Electricity Supply Company Limited (BESCOM) K.R. Circle, Bangalore - 560001	35. Managing Director Gulbarga Electricity Supply Company Limited (GESCOM) Station Main Road Gulburga (Karnataka) - 585102	36. Managing Director Hubli Electricity Supply Company Limited (HESCOM) Navanagar, PB Road Hubli (Karnataka) - 580025

37. Managing Director Mangalore Electricity Supply Company Limited (MESCOM) Paradigm Plaza, AB Shetty Circle Mangalore - 575001	38. Managing Director Chamundeshwari Electricity Supply Company Limited (CESCOM) # 927, L J Avenue, Ground Floor New Kantharajurs Road Saraswatipuram, Mysore - 570009	39. Chairman cum Managing Director Tamil Nadu Generation and Distribution Co. Ltd. (TANGEDCO 10 th floor, NPKRR Malligai, 144 Anna Salai, Chennai - 600002
 Superintending Engineer & HoD Electricity Department, Govt. of Puducherry Electricity Department 137, N.S.C. Bose Salai Puducherry - 605001 	41. Chairman & Managing Director Kerala State Electricity Board (KSEB) Vydyuthi Bhavanam, Pattom Thiruvananthapuram - 695004	42. Chief Engineer Department of Power, Govt. of Arunachal Pradesh Vidyut Bhawan, Zero-Point-Tin Itanagar (Ar.P.) - 791111
43. Chief General Manager Assam Electricity Grid Corporation Limited (AEGCL) 1st Floor, Bijulee Bhawan Paltan Bazar, Dist- Kamrup Guwahati (Assam) - 781001	44. Managing Director Manipur State Power Distribution Company Limited 3rd Floor, New Directorate Building Near 2nd M.L.Gate, Dimpur Road Imphal (Manipur) - 795001	45. Commissioner & Secretary Department of Power & Electricity Govt. of Mizoram Kawlphetha Building New Secretariat Complex Khatla, Aizawl (Mizoram) -796001
46. Superintending Engineer (T&G), Department of Power Govt. of Nagaland Electricity House, A.G. Colony Kohima (Nagaland) - 797005	47. Director (Distribution) Meghalaya Energy Corporation Limited (MeECL) Lum Jingshai, Short Round Road Shillong - 793001	48. AGM (Transmission) Tripura State Electricity Corporation Limited (TSECL) Transmission Circle Agartala, Tripura (West) - 799201

All State Transmission Utilities

 Chairman and Managing Director, Madhya Pradesh Power Transmission Co. Ltd. (MPPTCL) Block No.2, Shakti Bhawan, Rampur Jabalpur (MP) – 482 008. 	50. Director (Operation) Maharashtra State Electricity Transmission Co. Ltd. (MSETCL) Prakashganga, Plot no. C-19 Bandra Kurla Complex Bandra East, Mumbai - 400 051	51. Managing Director Gujarat Electricity Transmission Corporation Limited (GETCO) Sardar Patel Vidyut Bhawan Race Course, Baroda Gujarat - 390007
52. Executive Engineer Department of Electricity Administration of Daman & Diu Moti Daman, DD - 396220	53. Executive Engineer Electricity Department Transmission Division D & NH – Silvassa	54. Managing Director Chhattisgarh State Power Transmission Co. Ltd. (CSPTCL) Dangania, Raipur Chhattisgarh - 492013
55. Chief Electrical Engineer Govt of Goa Electricity Department 3rd floor, Vidyut Bhawan Panaji (Goa) - 403521	56. Director (W &P) UP Power Transmission Corporation Limited (UPPTCL) Shakti Bhawan Extn,3rd floor 14, Ashok Marg Lucknow (UP) - 226001	57. Director (Projects) Power Transmission Corporation of Uttarakhand Limited (PTCUL) Urja Bhawan Campus Kanawali Road Dehradun (UK) - 248001
58. Director (Technical), Punjab State Transmission Corporation Ltd. (PSTCL) Head Office, The Mall Patiala (Punjab) - 147001	59. Director (Operation) Delhi Transco Ltd. (DTL) Shakti Sadan Kotla Marg New Delhi - 110002	60. Director (Technical) Rajasthan Rajya Vidyut Prasaran Nigam Limited (RRVPNL) Vidut Bhawan Jaipur - 302005
61. Director (Technical) Haryana Vidyut Prasaran Nigam Limited (HVPNL) Shakti Bhawan, Sector-6 Panchkula - 134109	62. Managing Director, Himachal Pradesh Power Transmission Corp. Ltd. (HPPTCL) Barowalias, Khalini Shimla (H.P.) - 171002	63. Chief Engineer (Operation) Ministry of Power UT Secretariat Sector-9 D Chandigarh -161009
64. Development Commissioner (Power) Power Department Grid Substation Complex, Janipur Jammu (J&K) - 180007	65. PCE cum Secretary Energy & Power Department, Govt. of Sikkim Gangtok (Sikkim) - 737101	66. Director (Transmission), Orissa Power Transmission Corporation Limited (OPTCL) Jan path, Bhubaneshwar – 751022
67. Managing Director, Bihar State Power Transmission Company Ltd. (BSPTCL) Vidyut Bhawan, Bailey Road, Patna - 800021	68. Managing Director, Jharkhand Urja Sancharan Nigam Limited (JUSNL) Engineering Building, H.E.C. Dhurwa, Ranchi - 834004	69. Director (System Operation) West Bengal State Electricity Transmission Co. Ltd. (WBSETCL) Vidyut Bhawan, 5th Floor, Block-D Bidhannagar, Sector-II

		Kolkata – 700091
70. The Director (Grid & Trans. Mgt) Transmission Corp. of Andhra Pradesh Ltd. (APTRANSCO) Vidyut Soudha Hyderabad - 500082	71. The Director (Trans & System Op.) Kerala State Electricity Board Ltd, Vidyuthi Bhawanam Pattom, P.B. No. 1028 Thiruvananthapuram - 695004	72. Member (Distribution) Tamil Nadu Electricity Board(TNEB) 6th Floor, Eastern Wing 800 Anna Salai Chennai - 600002
73. Director (Transmission.) Transmission Corp. of Telangana Ltd. (TSTRANSCO) Vidyut Soudha, Khairatabad Hyderabad - 500082	74. Superintending Engineer-I Electricity Department First Floor, Gingy Salai Puducherry - 605 001	75. Director (Transmission) Karnataka Power Transmission Corporation Ltd. (KPTCL) Cauvery Bhawan, K.G. Road Bengaluru - 560009
76. Chief Engineer (Power) Department of Power Vidyut Bhawan, Zero Point Tinali Itanagar (Arunachal Pradesh)-791111	77. Managing Director, Assam Electricity Grid Corporation Limited (AEGCL) BijuleeBhawan; Paltan Bazar, Guwahati (Assam) - 781001	78. Chairman & Managing Director Meghalaya Energy Corporation Ltd. LumJingshai, Short Round Road Shillong (Meghalaya) - 793001
79. Chairman & Managing Director Tripura State Electricity Corporation Limited Bidyut Bhavan, Banamalipur, Agartala (Tripura) - 799201	80. Engineer-in-Chief Power & Electricity Department Govt. of Mizoram New Secretariat Complex Aizawl (Mizoram) - 796001	81. Chief Engineer (Power) Electricity Department Govt. of Manipur Keishampat Imphal (Manipur) - 795001
82. Chief Engineer Department of Power Govt. of Nagaland Electricity House, A.G. Colony Kohima (Nagaland) - 797005		

Others:

83. CEO POSOCO B-9, Qutab Institutional Area, Katwaria Sarai, New Delhi – 110010	84. Chairman and Managing Director POWERGRID Saudamini Plot no. 2, Sector-29, Gurugram - 122001	85. COO (CTU) POWERGRID Saudamini, Plot no. 2, Sector -29, Gurugram - 122001
86. Chairman and Managing Director North Eastern Electric Power Corporation Ltd. (NEEPCO) Brookland Compound Lower New Colony Shillong (Meghalaya) - 793003	87. Director (Projects) NTPC Limited NTPC Bhawan,Core-7 Scope Complex, Lodhi Road New Delhi - 110003 88. Director (T&RE) NHPC Limited NHPC Office Complex Sector-33, NHPC Faridabad (Haryana) - 1	
 Director (Technical) THDC India Limited Pragatipuram, Bypass Road Rishikesh (UK) - 249201 	90. Member (Power) Bhakra Beas Management Board Sectot-19 B, Madhya Marg Chandigarh - 160019	91. Director (Power), Neyveli Lignite Corp. Ltd. Corporate Office, Block-I Neyveli, Tamil Nadu - 607801
 92. Chief Engineer (Transmission) Nuclear Power Corporation of India Limited (NPCIL) 9-S-30, Vikram Sarabahai Bhawan, Anushakti Nagar, Mumbai - 400094 	3. Managing Director Adani Transmission Ltd. Adani House, Nr Mithakhali Six Roads, Navrangpura Ahmedabad (Gujarat) - 380009 94. Head-B.D. Sterlite Po Block C-2 "The Mira 1&2 Ishwa	94. Head-B.D. & Corporate Affairs Sterlite Power Limited Block C-2, 2nd Floor "The Mira Corporate Suites" 1&2 Ishwar Nagar, Mathura Road New Delhi - 110 065
95. Director Essel Infra Projects Limited A26/3, Mohan Co-operative Industrial Estate, Mathura Road New Delhi - 110044	96. Director (Operations) Kalpataru Power Transmission Ltd. 101, Kalpataru Synergy, Opp. Grand Hyatt, Vakola, Santacruz (E) Mumbai - 400055	97. CEO Power Finance Corporation Ltd. 'Urjanidhi',1, Barakhamba Lane Connaught Place New Delhi - 110001
98. CEO Rural Electrification Corporation Limited (REC) Core-4, Scope Complex 7, Lodhi Road New Delhi - 110003		

ST	A۸	IDA	١RD
----	----	-----	-----

IMPLEMENTATION AGREEMENT

FOR

DEVELOPMENT AND OPERATION OF INTER-STATE TRANSMISSION SYSTEM

FOR TRANSMISSION OF ELECTRICITY THROUGH TARIFF BASED COMPETITIVE BIDDING

BETWEEN THE

[Insert Name of Transmission Services Provider]

AND

[Insert Name of Upstream/Downstream Transmission System Developer or Associated Generating Station Developer]

.....2017

Draft Version 1.1 | May 2017

May 2017 V.1.1

IMPLEMENTATION AGREEMENT

This IMPLEMENTATION AGREEMENT (hereinafter referred to as "IA" or
"Agreement" or "the Agreement" or "this Agreement") is made on the[Insert
Day] of[Insert Month] Two Thousand and[Insert Year] at [Insert
Location].
Between:
[Insert Special Purpose Vehicle Name executing the
Project (SPV)], a company incorporated under the Companies Act, 2013, having its
registered office at[Insert Address], (hereinafter referred to as "TSP"
or "SPV" which expression shall unless repugnant to the context or meaning thereof
mean and include its successors and permitted assigns) as Party of the one part;
AND
[Insert Name of the Transmission System developer
developing Upstream or Downstream Transmission Project /Associated Generating
Station developer as applicable] a company incorporated under the Companies Act
[Insert relevant Act], having its registered office at[Insert Address]
(hereinafter referred to as "Developer" which expression shall unless repugnant to
the context or meaning thereof include its successors and permitted assigns) as
Party of the other part;

WHEREAS,

- A. The Ministry of Power, Government of India has notified the revised Standard Bidding Guidelines and as per the revised Bidding Guidelines an Implementation Agreement is to be executed by the SPV executing the Project and the Developer developing upstream or downstream transmission project and/or generating station associated with the Project.
- B. In accordance with the Bidding Guidelines, the Bid Process Coordinator (hereinafter referred to as BPC) had initiated a competitive e-reverse bidding process through issue of RFQ and RFP for selecting a Successful Bidder to build, own, operate and transfer the Project comprising of the Elements mentioned in Schedule 1.
- C. The BPC has incorporated the SPV to implement, operate and maintain the Transmission System/Project as per the license issued by CERC.
- D. The Implementation Agreement covers various modalities in case of delay in achieving COD either by the SPV or by the Developer. Further, the IA shall also include provisions for realignment of SCOD, and sharing of impact of delay in achieving SCOD by the Parties executing the IA.
- E. The terms and conditions stipulated in the Transmission License issued by the Commission in the name of SPV shall be applicable to this Agreement and the TSP agrees to comply with these terms and conditions.
- F. Further, the Successful Bidder of the Project shall execute Implementation and Service Agreement (ISA) with the Nodal Agency as per the Bidding Guidelines.
- G. In case of inconsistency between the License terms & conditions, conditions of ISA, and the conditions of this Agreement, the conditions stipulated in the License followed by ISA and then this Agreement in that order shall prevail.

NOW, THEREFORE, IN CONSIDERATION OF THE PREMISES AND MUTUAL AGREEMENTS, COVENANTS AND CONDITIONS SETFORTH HEREIN, IT IS HEREBY AGREED BY AND BETWEEN THE PARTIES HERETO AS FOLLOWS:

SECTION 1

1 DEFINITIONS AND INTERPRETATIONS

1.1 Definitions

1.1.1 The words / expressions used in this Agreement, unless as defined below or repugnant to the context, shall have the same meaning as assigned to them by the Electricity Act, 2003 and the rules or regulations framed there under including those issued / framed by the Commission (as defined hereunder), as amended or re-enacted from time to time or the General Clauses Act, failing which it shall bear its ordinary English meaning.

The words/expressions when used in this Agreement shall have the respective meanings as specified below:

"Act" or "Electricity Act" or "Electricity Act 2003" shall mean the Electricity Act, 2003 and any amendments made to the same or any succeeding enactment thereof;

"Adjudicator" shall mean as defined under Article 11.3 of this Agreement;

"Affiliate" shall mean a company that either directly or indirectly

- i. controls or
- ii. is controlled by or
- iii. is under common control with
- a Bidding Company (in the case of a single company) or a Member (in the case of a Consortium) and "control" means ownership by one entity of at least twenty six percent (26%) of the voting rights of the other company;

- "Annual Fixed Charges" shall mean the yearly fixed charges of a generating station as approved under Section 62 of the Act or adopted by the Appropriate Commission under Section 63 of the Act.;
- "Annual Transmission Charges" shall mean yearly transmission charges of a Transmission System as approved under Section 62 of the Act or adopted by the Appropriate Commission under Section 63 of the Act.;
- "Appropriate Commission" shall mean either the State Electricity Regulatory Commission or the Central Electricity Regulatory Commission under whose jurisdiction the Project falls.
- "Arbitration Tribunal" shall mean the tribunal constituted under Article 11.4.4 of this Agreement;
- "Associated Generating Station" shall mean generating station for which the Project has been identified by Ministry of Power, Government of India to be implemented for evacuation of power from such generating station.
- "Bidding Guidelines" shall mean the "Tariff Based Competitive Bidding Guidelines for Transmission Service" and "Guidelines for Encouraging Competition in Development of Transmission Projects" issued by Government of India, Ministry of Power dated 13th April 2006 under Section 63 of the Electricity Act and as amended from time to time;
- "Bid Process Coordinator" or "BPC" shall mean a person or its authorized representative as notified by the Government of India, responsible for carrying out the process for selection of Transmission Service Provider;
- "Business Day" shall mean a day other than Sunday or a statutory holiday, on which the banks remain open for business in the State in which the Nodal Agency's registered office is located and the concerned TSP is located;
- "CEA" shall mean the Central Electricity Authority constituted under Section -70 of the Electricity Act;

"Commercial Operation Date" or "COD" shall mean the date on which the Project has been declared under commercial operation;

Provided that the COD shall not be a date prior to the Scheduled COD as per Schedule 1, unless mutually agreed to by all Party(ies).

"Commission" or "CERC" shall mean the Central Electricity Regulatory Commission referred to in sub-section (1) of Section 76 of the Electricity Act, 2003 or its successors and assigns;

"Competent Court of Law" shall mean the Supreme Court or any High Court, or any tribunal or any similar judicial or quasi-judicial body in India that has jurisdiction to adjudicate upon issues relating to the Project;

"Consultation Period" shall mean the period of sixty (60) days or such longer period as the Parties may agree, commencing from the date of issue of a SPV's Preliminary Notice or a Developer's Preliminary Notice, as provided in Article 8.3.1 and 8.4.1 of this Agreement, for consultation between the Parties, to mitigate the consequence of the relevant event having regard to all the circumstances;

"Consents, Clearances and Permits" shall mean all authorizations, licenses, approvals, registrations, permits, waivers, privileges, acknowledgements, agreements, or concessions required to be obtained from or provided by any Indian Governmental Instrumentality for the development, execution and operation of Project including without any limitation for the construction, ownership, operation and maintenance of the Transmission Lines and/or sub-stations or any generating station;

"Construction Period" shall mean the period from (and including) the Effective Date of the Implementation and Service Agreement up to (but not including) the COD of the Element of the Project in relation to an Element and up to (but not including) the COD of the Project in relation to the Project;

"Contractors" shall mean the engineering, procurement, construction, operation & maintenance contractors, surveyors, advisors, consultants, designers, suppliers to the SPV or Developer and each of their respective sub-contractors (and each of their respective successors and permitted assigns) in their respective capacities as such;

"Central Transmission Utility" or "CTU" shall mean any government company, which the central government may notify under sub-section (1) of Section 38 of the Electricity Act 2003;

"Day" shall mean a day starting at 0000 hours and ending at 2400 hours;

"Developer's Preliminary Notice" shall mean a notice issued by the Developer pursuant to the provisions of Article 8.4.1 of this Agreement;

"Dispute" shall mean any dispute or difference of any kind between the SPV and the Developer, in connection with or arising out of this Agreement including any issue on the interpretation and scope of the terms of this Agreement as provided in Article 11.2.1;

"Downstream System" shall mean the immediately following transmission system(s) (being developed by a person) to the Project being developed by SPV and as identified by CEA.

"Effective Date" for the purposes of this Agreement, shall have the same meaning as per Article 2.1 of this Agreement;

"Element" or "Phase" of Transmission System shall mean each Transmission Line or each circuit of the Transmission Lines (where there are more than one circuit) or each bay of Sub-station or switching station or HVDC terminal or inverter station of the Project, including ICTs, Reactors, SVC, FSC, etc. forming part of the ISTS, which will be owned, operated and maintained by the concerned ISTS Licensee, and which has a separate Scheduled COD as per Schedule 1 of this Agreement and has a separate percentage for recovery of Transmission Charges on achieving COD;

"Event of Default" shall mean the events as defined in Article 8 of this Agreement;

"Expiry Date" shall be the date as per Schedule 1 of this Agreement;

"Financing Agreements" shall mean the agreements pursuant to which the SPV or Developer is to finance the Project including the loan agreements, security documents, notes, indentures, security agreements, letters of credit and other documents, as may be amended, modified, or replaced from time to time;

"Financial Year" shall mean a period of twelve months at midnight Indian Standard Time (IST) between 1st April & 31st March;

"Force Majeure" and "Force Majeure Event" shall have the meaning assigned thereto in Article 6;

"GOI" shall mean Government of India;

"Grid Code" / "IEGC" shall mean the Grid Code specified by the Central Commission under Clause (h) of sub-section (1) of Section 79 of the Electricity Act;

"Implementation and Service Agreement" or "ISA" shall mean agreement executed by Nodal Agency on behalf of DICs with the TSP in accordance with Standard Bidding Guidelines for development of ISTS;

"Indian Governmental Instrumentality" shall mean Government of India, Government of any State in India or any ministry, department, board, authority, agency, corporation, commission under the direct or indirect control of Government of India or any State Government or both, any political sub-division of any of them including any court or Commission or tribunal or judicial or quasi-judicial body in India but excluding the CTU, SPV, TSP and the Designated ISTS Customers;

"Interconnection Facilities" shall mean the facilities as may be set up for transmission of electricity through the use of the Project, on either one or

both side of generating station's / CTU's / STU's / ISTS Licensee's / Designated ISTS Customer's substations (as the case may be) which shall include, without limitation, all other transmission lines, gantries, sub-stations and associated equipments not forming part of the Project;

"ISTS Licensee" shall be the TSP or Developer as the case may be under this Agreement, consequent to having been awarded a Transmission License by the CERC and shall be referred to as the TSP or Developer or the ISTS Licensee, as the context may require in this Agreement;

"Law" or "Laws" in relation to this Agreement, shall mean all laws including electricity laws in force in India and any statute, ordinance, rule, regulation, notification, order or code, or any interpretation of any of them by an Indian Governmental Instrumentality having force of law and shall include all rules, regulations, decisions and orders of the Commission;

"Lenders" means the banks, financial institutions, multilateral funding agencies, non banking financial companies registered with the Reserve Bank of India (RBI), insurance companies registered with the Insurance Regulatory & Development Authority (IRDA), pension funds regulated by the Pension Fund Regulatory & Development Authority (PFRDA), mutual funds registered with Securities & Exchange Board of India (SEBI), etc., including their successors and assigns, who have agreed on or before COD of the Project to provide the SPV or the Developer with the debt financing described in the capital structure schedule, and any successor banks or financial institutions to whom their interests under the Financing Agreements may be transferred or assigned;

Provided that, such assignment or transfer shall not relieve any Party of its obligations to the other Party under this Agreement in any manner and shall also does not lead to an increase in the liability of either Party;

"Lenders' Representative" shall mean the person notified by the Lender(s) in writing as being the representative of the Lender(s) or the Security Trustee and such person as may from time to time be replaced by

the Lender(s) pursuant to the Financing Agreements by written notice to the TSP;

"Liquidated Damages" shall mean as defined under Article 4 of this Agreement;

"Month" shall mean a period of thirty (30) days from (and excluding) the date of the event;

"Monthly Fixed Charges" for any unit or generating station, after COD of the unit or station, shall mean the amount of Annual Fixed Charges for the relevant Contract Year divided by number of months of the year for which such tariff has been determined;

"Monthly Transmission Charges" for any Element(s) or of the Project shall mean the amount of Annual Transmission Charges for the relevant Contract Year divided by number of months of the year for which such tariff pertains to;

"Notification" shall mean any notification, issued in the Gazette of India:

"Project Assets" shall mean all physical and other assets relating to and forming part of the Project including:

- (a) rights over the Site for substations, ROW for transmission lines;
- (b) tangible assets such as civil works and equipment, plant and machinery including foundations, embankments, pavements, electrical systems,

communication systems, relief centres, administrative offices and Substations;

- (c) project facilities situated on the Site;
- (d) all rights of the Parties under the project agreements;
- (e) financial assets, such as receivables, security deposits etc;
- (f) insurance proceeds; and
- (g) Applicable Permits and authorisations relating to or in respect of the Transmission System or Generating Station in case the Developer is setting up a Generating Station;"

"Prudent Utility Practices" shall mean the practices, methods and standards that are generally accepted internationally from time to time by electric transmission utilities for the purpose of ensuring the safe, efficient and economic design, construction, commissioning, operation, repair and maintenance of the Project and which practices, methods and standards shall be adjusted as necessary, to take account of:

- (i) operation, repair and maintenance guidelines given by the manufacturers to be incorporated in the Project,
- (ii) the requirements of Law, and
- (iii) the physical conditions at the Site;
- (iv) the safety of operating personnel and human beings;

"RLDC" shall mean the relevant Regional Load Dispatch Centre as defined in the Electricity Act, 2003, in the region(s) in which the Project is located;

"Scheduled COD" or "SCOD" in relation to an Element(s) shall mean the date(s) as mentioned in Schedule 1 as against such Element(s) and in relation to the Project, shall mean the date as mentioned in Schedule 1 as against such Project, or such date as may be mutually agreed among the Parties as per this Agreement;

- "Selected Bid" shall mean the Non-Financial Bid and the Final Offer of the Selected Bidder submitted during e-reverse bidding;
- "Sharing Regulations" shall mean the Central Electricity Regulatory Commission (Sharing of Inter-State Transmission Charges and Losses) Regulations, 2010 and as amended from time to time;
- "Site" in relation to a substation, switching station or HVDC terminal or inverter station, shall mean the land and other places upon which such station / terminal is to be established;
- "STU" or "State Transmission Utility" shall be the board or the Government company, specified as such by the State Government under sub-section (1) of Section 39 of the Electricity Act 2003;
- "SPV's Preliminary Notice" shall mean a notice issued by the holding company of SPV pursuant to the provisions of Article 8.3.1 of this Agreement;
- "Term of Agreement" for the purposes of this Agreement shall have the meaning ascribed thereto in Article 2.2 of this Agreement;
- "Transmission Charges" shall mean the Final Offer of the Selected Bidder during the e-reverse bidding and adopted by the Commission, payable to the SPV by the DICs, as per the provisions of ISA.
- "Transmission License" shall mean the license granted by the Commission in terms of the relevant regulations for grant of such license issued under the Electricity Act;

"Transmission Lines" shall mean all high pressure cables and overhead lines (not being an essential part of the distribution system of a licensee) transmitting electricity from a generating station to another generating station or a sub-station, together with any step-up and step-down transformers, switch-gear and other works necessary to and used for the control of such cables or overhead lines, and such buildings or part thereof as may be required to accommodate such transformers, switchgear and other works;

"Transmission System/Project" shall mean [Insert Names of the Projects along with the SPV and Developer], as detailed in Schedule 1 of this Agreement;

"Transmission Service Agreement" or "TSA" shall be the agreement executed between the relevant parties pursuant to Regulation 2 (u) of the Central Electricity Regulatory Commission (Sharing of Inter State Transmission Charges and Losses) Regulation 2010 and as amended from time to time;

"Transmission System" shall mean a line with associated sub-stations or a group of lines inter-connected together along with associated sub-stations and the term includes equipment associated with transmission lines and sub-stations;

"**Upstream System**" shall mean the immediately preceding transmission system(s) (being developed by a person) to the Project being developed by SPV and as identified by CEA.

1.2 Interpretation:

Save where the contrary is indicated, any reference in this Agreement to:

"Agreement" shall be construed as including a reference to its Schedules, Appendices and Annexures;

"Rupee", "Rupees" and "Rs." shall denote lawful currency of India;

"crore" shall mean a reference to ten million (10,000,000) and a "lakh" shall mean a reference to one tenth of a million (1,00,000);

"encumbrance" shall be construed as a reference to a mortgage, charge, pledge, lien or other encumbrance securing any obligation of any person or any other type of preferential arrangement (including, without limitation, title transfer and retention arrangements) having a similar effect;

"indebtedness" shall be construed so as to include any obligation (whether incurred as principal or surety) for the payment or repayment of money, whether in present or future, actual or contingent;

"person" shall have the meaning as defined in Section 2 (49) of the Act;

"winding-up", "dissolution", "insolvency", or "reorganization" in the context of a company or corporation shall have the same meaning as defined in the Companies Act, 1956/ Companies Act, 2013 (as the case may be).

- 1.2.1 Words importing the singular shall include the plural and vice versa.
- 1.2.2 This Agreement itself or any other agreement or document shall be construed as a reference to this or to such other agreement or document as it may have been, or may from time to time be, amended, varied, novated, replaced or supplemented.
- 1.2.3 A Law shall be construed as a reference to such Law including its amendments or re-enactments from time to time.
- 1.2.4 A time of day shall, save as otherwise provided in any agreement or document be construed as a reference to Indian Standard Time.
- 1.2.5 Different parts of this Agreement are to be taken as mutually explanatory and supplementary to each other and if there is any inconsistency between or among the parts of this Agreement, they shall be interpreted in a harmonious manner so as to give effect to each part.
- 1.2.6 The tables of contents and any headings or sub-headings in this Agreement have been inserted for ease of reference only and shall not affect the interpretation of this Agreement.
- 1.2.7 All interest payable under this Agreement shall accrue from day to day and be calculated on the basis of a year of three hundred and sixty five (365) days.

1.2.8 The words "hereof" or "herein", if and when used in this Agreement shall mean a reference to this Agreement.

2 EFFECTIVENESS AND TERM OF THE AGREEMENT

2.1 Effective Date

2.1.1 This Agreement shall be effective from the date of Transfer of SPV to the Selected Bidder.

2.2 Term of the Agreement

This Agreement shall continue to be effective, in relation to each Phase, until the Expiry Date as per Article 2.3, when it shall automatically terminate.

2.3 Expiry Date

The Agreement shall expire once the Transmission System/Project of SPV and Developer have achieved COD as per Schedule 1 of this Agreement.

2.4 Survival

2.4.1 The expiry of this Agreement shall not affect any accrued rights, obligations and liabilities of the Parties under this Agreement including but not limited to those specifically provided in this Agreement nor shall it affect any continuing obligations which this Agreement provides either expressly or by necessary implication, which are to survive its expiry.

3 DEVELOPMENT AND CONSTRUCTION OF PROJECT BY THE PARTIES

3.1 SPV's obligations in development and construction of the Transmission Project

- 3.1.1 Subject to the terms and conditions of this Agreement, SPV at its own cost and expense shall observe, comply with, perform, undertake and be responsible;
 - (a) To comply with the provisions of this Agreement
 - (b) To arrange for funds and build, own, operate and commission the Project as per the detailed scope of work set out in Schedule 1 of this Agreement and achieve COD not later than SCOD.
 - (c) To procure and maintain in full force and effect the Consents, Clearances and Permits required by it pursuant to this Agreement and in accordance with Law.
 - (d) To submit quarterly status report towards planned and actual work execution in the Agreed Form to CEA and Nodal Agency.
 - (e) To adhere to the safety standards prescribed by various authorities.

3.2 Developer's obligations in development and construction of its project

- 3.2.1 Subject to the terms and conditions of this Agreement, Developer at its own cost and expense shall observe, comply with, perform, undertake and be responsible;
 - (a) To comply with the provisions of this Agreement
 - (b) To arrange for funds and build, own, operate and commission the project as per the detailed scope of work and achieve COD not later than SCOD.
 - (c) To procure and maintain in full force and effect the Consents, Clearances and Permits required by it pursuant to this Agreement and in accordance with Law.
 - (d) To submit quarterly status report towards actual work execution in the Agreed Form to CEA and Nodal Agency.
 - (e) To adhere to the Safety Standards prescribed by various authorities.

3.3 Commencement and Completion Time

- 3.3.1 The Parties shall take all necessary steps to commence work on their respective projects and proceed with their implementation in order to achieve COD by SCOD as per Schedule 1 of this Agreement.
- 3.3.2 The Parties may also mutually agree for early commissioning of their projects provided that the SPV is allowed early commissioning of the Project by competent authority.
- 3.3.3 Under no circumstances shall the SCOD be postponed unless expressly provided for under Article 3.4 of this Agreement.

3.4 Extension of Time

- 3.4.1 In the event that an Element or the Project cannot be commissioned by its Scheduled COD on account of any Force Majeure Event as per Article 6, the Scheduled COD shall be extended, by a 'day for day' basis, for a maximum period of one hundred and eighty (180) days.
- 3.4.2 If the Parties have not agreed, within thirty (30) days after the Affected Party's performance has ceased to be affected by the relevant circumstance, on how long the Scheduled COD should be deferred by, any Party may raise the Dispute to be resolved in accordance with Article 11.

4 Liquidated Damages

4.1 Liquidated Damages Payable due to delay on account of SPV

- 4.1.1 If SPV fails to achieve COD of the Project or any of the Phase of the Project by SCOD and the Developer has achieved COD or is ready to achieve COD but is unable to do so due to SPV's failure to achieve its COD by SCOD for the reasons other than Force Majeure, then the SPV subject to Article 4.1.2 shall pay to the Developer, a sum equivalent to 3.33% of Monthly Transmission Charges applicable for the Element of the Project of the SPV [in case where no Elements have been defined, on the Project as a whole] / Project, for each day of delay as Liquidated Damages for such delay and not as penalty, without prejudice to any rights of the Nodal Agency under the ISA.
- 4.1.2 In case the SPV has executed IA with only one Developer, the SPV shall pay the Liquidated Damages collected as per Article 4.1.1 as per the following principles;
 - (a) In case the Developer is implementing transmission project under TBCB route and has achieved COD or is ready to achieve COD but is unable to do so due to SPV's failure to achieve COD, the Liquidated Damages shall be payable to the Developer equivalent to
 - Minimum of the 3.33% of the Monthly Transmission Charge applicable for the Element of the Project that the Developer is implementing or the Element of the Project that SPV is implementing [in case where no Elements have been defined, to be on the Project as a whole] / Project, for each day of delay as liquidated damages. Such liquidated damages shall be paid for only such Element(s) of the Project that has been affected due to SPV's failure to achieve COD within the timelines specified.

Provided that in case of various elements forming part of the Project, and not all Elements have been affected due to delay, the SPV shall pro-rata allocate the Liquidated Damages on the these element(s) of the Project based on the percentage of quoted transmission charges provided at the time of bidding.

- 4.1.3 In case the Developer is implementing transmission project other than TBCB route, the SPV shall pay on the Liquidated Damages to the Developer for such affected Elements of the Project that have been affected due to SPV's failure to achieve COD in the manner and amount as computed under Article 4.1.2(a) upon such impact being approved by Appropriate Commission.
- 4.1.4 In case the SPV's Project is associated with evacuation of power from associated generating station being set up under TBCB Route and SPV fails to achieve COD by SCOD and the generating station has achieved COD or is unable to achieve COD due to SPV's failure to achieve COD, and the SPV is unable to make alternative arrangements for evacuation of such power, the SPV shall pay the Liquidated Damages to the Developer which would be minimum of the following;
 - (a) 3.33% of monthly fixed charges applicable for the generating station, for each day of delay as Liquidated Damages for such delay and not as penalty; or
 - (b) The amount arrived and payable on the basis of Article 4.1.1 of this Agreement.

- 4.1.5 In case the SPV's Project is associated with evacuation of power from associated generating station being set up other than TBCB Route and SPV fails to achieve COD by SCOD and the generating unit/station has achieved COD or is unable to achieve COD due to SPV's failure to achieve COD, and the SPV is unable to make alternative arrangements for evacuation of such power, the Nodal Agency shall pay the Liquidated Damages collected from SPV to the Developer which would be minimum of the following;
 - (a) 3.33% of monthly fixed charges of the generating unit/station as approved by the Appropriate Commission, for each day to the generating station as Liquidated Damages; or
 - (b) The amount arrived and payable on the basis of Article 4.1.1 of this Agreement.
- 4.1.6 The SPV's maximum liability under this Article 4.1 shall be limited to the amount of Liquidated Damages calculated in accordance with Article 4.1.1 and 4.1.2 for and up to six (6) months of delay for the Element or the Project.
- 4.1.7 In case two or more Implementation Agreement are executed by the SPV, and due to failure of SPV to achieve COD by SCOD, two or more Projects who have either achieved COD or are unable to achieve COD due to SPV failure to achieve COD, the amount payable by SPV as per Article 4.1.1 shall be allocated among various affected projects as per following Principles;
 - (a) The Liquidated Damages shall be pro-rata allocated among various projects on the basis of their Annual Transmission Charges for the affected Element/Project for the first year in case of transmission projects and Annual Fixed Charge for the first year corresponding to generating unit/station affected in case of associated generating stations.
 - (b) The Transmission Charges for Projects under TBCB route shall be equal to the total Transmission charges as per the bid and as adopted by the Commission for the first year. For Projects being developed other than through the TBCB route, the transmission charges shall be equal to as approved by the Appropriate Commission for the First Year.
 - (c) In case of Associated Generating Station developed under TBCB route the Annual Fixed Charges as per bid and as adopted by the Appropriate

Commission for the first year shall be considered. In case the generating station is being developed other than through the TBCB route, the Annual Fixed Charges as determined by the Appropriate Commission for the first year shall be considered.

Provided that in case the SPV arranges for alternative option for evacuation of power from Associated Generating Station, Liquidated Damages shall not be payable to such Developer and the same shall not be considered for computing pro-rata Liquidated Damages to other Developers.

4.2 Liquidated Damages Payable due to delay on account of Developer

- 4.2.1 In case the SPV has executed IA with a Developer developing transmission project and the Developer has failed to achieve COD by SCOD due to which SPV is unable to achieve COD or has achieved COD by some alternate means, the Developer shall pay the SPV Liquidated Damages for each day of delay in the manner and amount as per Article 4.1.1 in case the Developer is developing Transmission System/Project other than TBCB route and as per Article 4.1.2 in case the Developer is developing Transmission System/Project under TBCB route.
- 4.2.2 In case the SPV has executed IA with a Developer developing generating station and the SPV is unable to achieve COD due to failure on the part of Developer to achieve COD by SCOD, or has achieved COD due to alternative means, the Developer shall pay the SPV Liquidated Damages for each day of delay in the manner and amount as per Article 4.1.1 of this Agreement.
- 4.2.3 In case the SPV is unable to achieve COD due to failure on the part of two or more Developers with whom SPV has executed the IA, the SPV shall collect and Developer shall pay Liquidated Damages in the manner as per Article 4.1.1 and 4.1.2 in proportion to the first year of Annual Transmission Charges in case the Developer is involved in development of transmission project and Annual Fixed Charges in case the Developer is developing generating station as the case may be.
- 4.2.4 The maximum Liquidated Damages payable except in case of Article 4.2.2 to the SPV shall not exceed the amount as determined in accordance with Article 4.1.6 of this Agreement.

5 Commissioning and Commercial Operation of the Project

5.1 Commissioning of the Project

- 5.1.1 Either Party shall give the other Party at least sixty (60) days advance written notice of the date on which it intends to commission a Phase of the Project which shall not be a date earlier than the SCOD agreed by the Parties unless the same has been mutually agreed or allowed under this Agreement.
- 5.1.2 Either Party shall notify the other Party as soon as it believes that any Phase of the Project has been completed and ready to be connected with the Interconnection Facilities in accordance with this Agreement.

5.2 Commercial Operation

- 5.2.1 A Phase of the Project shall be declared Commissioned and the Commercial Operation Date (COD) shall be achieved upon fulfilment of following conditions;
 - (a) In case of Transmission Project being developed under TBCB mode the conditions for COD as laid out in the ISA have been fulfilled.
 - (b) In case of Transmission Project being developed under any other mode except for TBCB the conditions for COD as laid out by the State/Central Commission have been fulfilled.
 - (c) In case of Generating Station, on acceptance of COD by CEA.

6 Force Majeure

An "Affected Party" means any of the Parties to this Agreement whose performance has been adversely affected by an event of Force Majeure.

A 'Force Majeure' means any event or circumstance or combination of events and circumstances including those stated below that wholly or partly prevents or unavoidably delays an Affected Party in the performance of its obligations under this Agreement, but only if and to the extent that such events or circumstances are not within the reasonable control, directly or indirectly, of the Affected Party and could not have been avoided if the Affected Party had taken reasonable care or complied with Prudent Utility Practices:

6.1 Natural Force Majeure Events:

Act of God, including, but not limited to drought, fire and explosion (to the extent originating from a source external to the Site), earthquake, volcanic eruption, landslide, flood, cyclone, typhoon, tornado, or exceptionally adverse weather conditions, which are in excess of the statistical measures for the last hundred (100) years;

6.2 Non-Natural Force Majeure Events:

6.2.1 Direct Non-Natural Force Majeure Events

- (a) Nationalization or compulsory acquisition by any Indian Governmental Instrumentality of any material assets or rights of the Affected Party; or
- (b) The unlawful, unreasonable or discriminatory revocation of, or refusal to renew, any Consents, Clearances and Permits required by the Affected Party to perform their obligations under this Agreement or any unlawful, unreasonable or discriminatory refusal to grant any other Consents, Clearances and Permits required for the development/ operation of the Project, provided that a Competent Court of Law declares the revocation or refusal to be unlawful, unreasonable and discriminatory and strikes the same down; or
- (c) Any other unlawful, unreasonable or discriminatory action on the part of an Indian Governmental Instrumentality which is directed against the Project,

provided that a Competent Court of Law declares the action to be unlawful, unreasonable and discriminatory and strikes the same down.

6.2.2 Indirect Non - Natural Force Majeure Events

- (a) Act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo, revolution, riot, insurrection, terrorist or military action; or
- (b) Radioactive contamination or ionising radiation originating from a source in India or resulting from any other Indirect Non Natural Force Majeure Event mentioned above, excluding circumstances where the source or cause of contamination or radiation is brought or has been brought into or near the Site by the Affected Party or those employed or engaged by the Affected Party; or
- (c) Industry-wide strikes and labour disturbances, having a nationwide impact in India.

6.3 Force Majeure Exclusions

- 6.3.1 Force Majeure shall not include (i) any event or circumstance which is within the reasonable control of the Parties and (ii) the following conditions, except to the extent that they are consequences of an event of Force Majeure:
 - (a) Unavailability, late delivery, or changes in cost of the machinery, equipment, materials, spare parts etc. for the Project;
 - (b) Delay in the performance of any Contractors or their agents;
 - (c) Non-performance resulting from normal wear and tear typically experienced in such projects;
 - (d) Strikes or labour disturbance at the facilities of the Affected Party;
 - (e) Insufficiency of finances or funds or the Agreement becoming onerous to perform; and
 - (f) Non-performance caused by, or connected with, the Affected Party's:
 - (i) negligent or intentional acts, errors or omissions;

- (ii) failure to comply with an Indian Law; or
- (iii) breach of, or default under this Agreement or any Project Documents.

6.4 Notification of Force Majeure Event

6.4.1 The Affected Party shall give notice to the other Party of any event of Force Majeure as soon as reasonably practicable, but not later than seven (7) days after the date on which such Party knew or should reasonably have known of the commencement of the event of Force Majeure. If an event of Force Majeure results in a breakdown of communications rendering it unreasonable to give notice within the applicable time limit specified herein, then the Party claiming Force Majeure shall give such notice as soon as reasonably practicable after reinstatement of communications, but not later than one (1) day after such reinstatement.

Provided that, such notice shall be a pre-condition to the Affected Party's entitlement to claim relief under this Agreement. Such notice shall include full particulars of the event of Force Majeure, its effects on the Party claiming relief and the remedial measures proposed. The Affected Party shall give the other Party monthly reports on the progress of those remedial measures and such other information as the other Party may reasonably request about the Force Majeure.

6.4.2 The Affected Party shall give notice to the other Party of (i) the cessation of the relevant event of Force Majeure; and (ii) the cessation of the effects of such event of Force Majeure on the performance of its rights or obligations under this Agreement, as soon as practicable after becoming aware of each of these cessations.

6.5 Duty to perform and duty to mitigate

To the extent not prevented by a Force Majeure Event, the Affected Party shall continue to perform its obligations as provided in this Agreement. The Affected Party shall use its reasonable efforts to mitigate the effect of any event of Force Majeure as soon as practicable.

6.6 Available Relief for a Force Majeure Event

Subject to this Article 6,

- (a) No Party shall be in breach of its obligations pursuant to this Agreement to the extent that the performance of its obligations was prevented, hindered or delayed due to a Force Majeure Event;
- (b) Each Party shall be entitled to claim relief for a Force Majeure Event affecting its performance in relation to its obligations under Articles 3.1and 3.2, of this Agreement.

For so long as the Affected Party is claiming relief due to any Force Majeure Event under this Agreement, the other Party may, from time to time on one (1) day notice, inspect the Project and the SPV shall provide the other Party's personnel including its representatives with access to the Project to carry out such inspections, subject to the other Party's personnel complying with all reasonable safety precautions and standard.

6.7 Remedy to Continuation of Force Majeure

- 6.7.1 In case Parties could not reach an agreement pursuant to Article 3.4 of this Agreement and the Force Majeure Event or its effects continue to be present, either Party may approach CERC for further course of action as per the Electricity Act, 2003.
- 6.7.2 Notwithstanding any other provision of this Agreement, an SPV Preliminary Notice, Developer Preliminary Notice, or a notice under Article 13.5 shall be deemed to have been validly served on the service of any such notice.

7 Representation and Warranties

7.1 Representations and Warranties of the SPV

SPV hereby represents and warrants to and agrees with the Developer as follows and acknowledges and confirms that Developer is relying on such representations and warranties in connection with the transactions described in this Agreement:

- a) It has all requisite power to enter into and to execute and consummate this Agreement;
- b) This Agreement is enforceable against it in accordance with its terms;
- c) The consummation of the transactions contemplated by this Agreement on the part of SPV will not violate any provision of nor constitute a default under, nor give rise to a power to cancel any charter, mortgage, deed of trust or lien, lease, agreement, license, permit, evidence of indebtedness, restriction, or other contract to which SPV is a Party or to which SPV is bound, which violation, default or power has not been waived;
- d) SPV is not insolvent and no insolvency proceedings have been instituted, nor threatened or pending by or against SPV;
- e) There are no actions, suits, claims, proceedings or investigations pending or, to the best of SPVs knowledge, threatened in writing against SPV at law, in equity, or otherwise, and whether civil or criminal in nature, before or by, any court, commission, arbitrator or governmental agency-or authority, and there are no outstanding judgments, decrees or orders of any such courts, commission, arbitrator or governmental agencies or authorities, which materially adversely affect its ability to execute the Transmission System/ Project or to comply with its obligations under this Agreement.
- f) The evaluation report(s), specifications, drawings and other technical documents as provided to the Developer pursuant to this Agreement, are in accordance with Prudent Utility Practices.

7.2 Representations and Warranties of the Developer

The Developer hereby represents and warrants to and agrees with SPV as follows and acknowledges and confirms that SPV is relying on such representations and warranties in connections with the-transactions described in this Agreement;

- (a) It has all requisite power authorizing and has been duly authorized to execute and consummate this Agreement;
- (b) This Agreement is enforceable against it in accordance with its terms;
- (c) The consummation of the transactions contemplated by this Agreement on the part of the Developer will not violate any provision of nor constitute a default under, nor give rise to a power to cancel any charter, mortgage, deed of trust or lien, lease, agreement, license, permit, evidence of indebtedness, restriction, or other contract to which the Developer is a Party or to which the Developer is bound which violation, default or power has not been waived;
- (d) The Developer is not insolvent and no insolvency proceedings have been instituted, nor threatened or pending by or against the Developer;
- (e) There are no actions, suits, claims, proceedings or investigations pending or, to the best of Developer's knowledge, threatened in writing against the Developer at law, in equity, or otherwise, and whether civil or criminal in nature, before or by, any court, Commission, arbitrator or governmental agency or authority, and there are no outstanding judgments, decrees or orders of any such courts, commission, arbitrator or governmental agency or authority, which materially adversely affect its ability to execute the project or to comply with its obligations under this Agreement.
- 7.2.1 Both Parties agree and make all the representations and warranties above to be valid as on the Effective Date of this Agreement.

8 Events of Default

8.1 Developer's Events of Default

The occurrence and continuation of any of the following events shall constitute a Developer's Event of default, unless any such Event of Default occurs as a result of a breach by SPV of its obligation under this Agreement, a SPV's Event of Default, or a Force Majeure Event;

(a) After having taken up the construction of the Transmission System/Project, the abandonment by the Developer or the Developer's Contractors of the construction of the Transmission System/Project provided that the Developer and the Developer's contractors shall not be deemed to have abandoned construction of the Transmission System/Project if and for so long as it is using its diligent efforts to restart such construction;

(b) If the Developer:

- (i) assigns, mortgages or charges or purports to assign, mortgage or charge any of its assets or rights in contravention of the provision of this Agreement; or
- (ii) transfers or novates any of its obligations pursuant to this Agreement, in a manner contrary to the provisions of this Agreement;
- (c) The transfer, pursuant to law, of either:
 - (i) the rights and/or obligations of the Developer under, this Agreement; or
 - (ii) all or a substantial portion of the assets or undertakings of the Developer, except where such transfer;
 - (1) does not affect the ability of the transferee to perform, and such transferee has the financial and technical capability to perform, its obligations under this Agreement;
 - (2) is to a transferee who assumes such obligations under the Transmission System / Project and this Agreement remains effective with respect to the transferee.
- (d) The dissolution, pursuant to law, of the Developer except for the purpose of a merger, consolidation or re-organization that does not affect the ability of the resulting entity to perform its obligations under this Agreement and provided that such resulting entity expressly assumes such obligations;

- (e) Revocation of Transmission License of Developer;
- (f) In case the Developer is setting up a transmission system under TBCB mode, SPV shall have the Right to Terminate this Agreement in case the Nodal Agency has terminated the ISA executed with the Developer or the Commission has rejected/revoked the Transmission License of the Developer.
- (g) In case the Developer is setting up a transmission system under any other mode except TBCB, SPV shall have the Right to Terminate this Agreement in case the approval to execute such project has been withdrawn by the approving authority.
- (h) In case the Developer is setting up a generating station, SPV shall have the Right to Terminate this Agreement in case the approval for the project has been withdrawn by the competent authority.

8.2 SPV's Events of Default

The occurrence and continuation of any of the following events shall constitute SPV's Event of Default, unless any such SPV's Event of Default occurs as a result of breach by the Developer of its obligations under this Agreement, a Developer's Event of Default, or a Force Majeure Event;

- (a) After having taken up the construction of the Transmission System/Project, the abandonment by the SPV or the SPV's Contractors of the construction of the Transmission System/Project provided that the SPV and the SPV's contractors shall not be deemed to have abandoned construction of the Transmission System/Project if and for so long as it is using its diligent efforts to restart such construction;
- (b) The dissolution, pursuant to Law, of SPV of all or a substantial portion of the assets, unless such dissolution is in connection with the reorganization or re-structuring of the power sector and provided that (i) SPV's successor under this Agreement expressly assumes the obligations of SPV under this Agreement; or (ii) SPV's successor is either an owner or operator of a substantial portion of the transmission system for India; or (iii) SPV's

- successor is otherwise capable of performing the obligations of SPV under this Agreement;
- (c) The transfer pursuant to Law of either the rights and/or obligations of SPV hereunder or all or a substantial of the assets or undertakings of SPV, provided such transferee agrees to be bound by terms and conditions of this Agreement.
- (d) Revocation of Transmission License of the SPV;
- (e) The Nodal Agency has terminated the ISA executed with the TSP or the Commission has rejected/revoked the Transmission License of the Developer.
- (f) If SPV transfers or novates any of its obligations pursuant to this Agreement, in a manner contrary to the provisions of this Agreement.

8.3 Procedure upon Developer Events of Default

- 8.3.1 Upon the occurrence of a Developer's Event of Default, SPV may serve notice on Developer of its intention to terminate this Agreement (an "SPV Preliminary Notice") which notice shall specify in reasonable detail, the circumstances giving rise to such SPV Preliminary Notice.
- 8.3.2 Following the giving of an SPV Preliminary Notice, the Consultation Period shall apply during which the Parties shall consult for a period of up to two (2) months (or such, longer period as they may agree) (the "Notice Period") as to what steps shall be taken with a view to mitigating the consequences of the relevant event having regard to all the circumstances.
 - Provided further that during such period, the Developer shall not remove any equipment or any part of the Transmission System/ Project, without the prior written consent of SPV.
- 8.3.3 During any Notice Period, both Parties shall, save as otherwise provided in this Agreement, continue to perform their respective obligations under this Agreement.

8.3.4 Following the expiry of the Notice Period and unless the Parties shall have otherwise agreed or the circumstances giving rise to SPV Preliminary Notice shall have ceased to exist or shall have been remedied, SPV may approach CERC for further course of action as per the Electricity Act, 2003.

8.4 Procedure upon SPV Events of Default

- 8.4.1 Upon the occurrence of an SPV's Event of default, the Developer may serve notice on SPV of its intention to terminate this Agreement (a "Developer Preliminary Notice") which notice shall specify in reasonable detail the circumstance giving rise to such Developer Preliminary Notice.
- 8.4.2 Following the giving of a Developer Preliminary Notice, the Consultation Period shall apply during which the Parties shall consult for a period of up to two (2) months (or such longer period as they may agree) (the "Notice Period") as to what steps shall be taken with a view to mitigating the consequences of the relevant event having regard to all the circumstances,
- 8.4.3 During any Notice Period, both Parties shall, save as otherwise provided in this Agreement, continue to perform their respective obligations under this Agreement.
- 8.4.4 Following the expiry of the Notice Period and unless the Parties shall have otherwise agreed or the circumstances giving rise to the Developer Preliminary Notice shall have ceased to exist or shall have been remedied, Developer may approach CERC for further course of action as per the Electricity Act, 2003.

9 Liability and Indemnification

9.1 Limitation of Liability

- 9.1.1 Notwithstanding any other term or provision of this Agreement, the aggregate liability which each of the Parties shall have for all claims made under this Article shall not exceed the limit fixed as per the terms of this Agreement.
- 9.1.2 Except as expressly provided in this Agreement, neither the SPV nor the Developer nor their respective officers, directors, agents, employees or Affiliates (including, officers, directors, agents or employees of such Affiliates), shall be liable or responsible to the other Party or its Affiliates including its officers, directors, agents, employees, successors, insurers or permitted assigns for incidental, indirect or consequential, punitive or exemplary damages, connected with or resulting from performance or non-performance of this Agreement, or anything done in connection herewith, including claims in the nature of lost revenues, income or profits (other than payments expressly required and properly due under this Agreement), any increased expense of, reduction in or loss of transmission capacity or equipment used therefor, irrespective of whether such claims are based upon breach of warranty, tort (including negligence, whether of the SPV, the Developer or others), strict liability, contract, breach of statutory duty, operation of law or otherwise.
- 9.1.3 The SPV shall have no recourse against any officer, director or shareholder of the Developer or any Affiliate of the Developer or any of its officers, directors or shareholders for such claims excluded under this Article. The Developer shall also have no recourse against any officer, director or shareholder of the SPV, or any Affiliate of the SPV or Nodal Agency or any of its officers, directors or shareholders for such claims excluded under this Article.

9.2 Indemnification by SPV

- 9.2.1 The SPV shall indemnify, defend and hold the Developer harmless against:
 - (a) any and all third party claims, actions, suits or proceedings against the Developer for any loss of or damage to property of such third party, or death or injury to such third party, arising out of a breach by the SPV of any of its obligations under this Agreement, except to the extent that any

- such claim, action, suit or proceeding has arisen due to a negligent act or omission, breach of this Agreement or non-fulfilment of statutory duty on the part of Developer; and
- (b) any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest actually suffered or incurred by the Developer from third party claims arising by reason of:
 - (i) a breach by the SPV of any of its obligations under this Agreement, (provided that this Article shall not apply to such breaches by the SPV, for which specific remedies have been provided for under this Agreement) except to the extent that any such losses, damages, costs and expenses including legal costs, fines, penalties and interest (together to constitute "Indemnifiable Losses") has arisen due to a negligent act or omission, breach of this Agreement or non-fulfilment of statutory duty on the part of the Developer, or
 - (ii) any of the representations and warranties of the SPV under this Agreement being found to be inaccurate or untrue.
- 9.2.2 The Developer shall indemnify, defend and hold the SPV harmless against:
 - (a) any and all third party claims, actions, suits or proceedings against the SPV, for any loss of or damage to property of such third party, or death or injury to such third party, arising out of any material breach by the Developer of any of their obligations under this Agreement, except to the extent that any such claim, action, suit or proceeding has arisen due to a negligent act or omission, breach of this Agreement or breach of statutory duty on the part of the SPV, its Contractors, servants or agents; and
 - (b) any and all losses, damages, costs and expenses including legal costs, fines, penalties and interest ('Indemnifiable Losses') actually suffered or incurred by the SPV from third party claims arising by reason of:
 - (c) any material breach by the Nodal Agency of any of its obligations under this Agreement (provided that, this Article 9 shall not apply to such breaches by the Developer, for which specific remedies have been provided for under this Agreement), except to the extent that any such Indemnifiable Losses have

- arisen due to a negligent act or omission, breach of this Agreement or breach of statutory duty on the part of the SPV, its Contractors, servants or agents or
- (d) any of the representations and warranties of the Developer under this Agreement being found to be inaccurate or untrue.

9.3 Patent Indemnity

9.3.1 The Developer shall, subject to the SPV's compliance with Article 9.3.2, indemnify and hold harmless the SPV and its employees and officers from and against any and all suits, actions or administrative proceedings, claims, demands, losses, damages, costs, and expenses of whatsoever nature, including attorney's fees and expenses, which the SPV may suffer as a result of any infringement or alleged infringement of any patent, utility model, registered design, trademark, copyright or other intellectual property right registered or otherwise existing at the date of the Agreement by reason of the setting up of the Project by the Developer.

Such indemnity shall not cover any use of the Project or any part thereof other than for the purpose indicated by or to be reasonably inferred from the Agreement, any infringement resulting from the misuse of the Project or any part thereof, or any products produced in association or combination with any other equipment, plant or materials not supplied by the SPV, pursuant to the Agreement.

9.3.2 The SPV, subject to the Developer's compliance with Article 9.3.1 shall indemnify and hold harmless the SPV and its employees, officers from and against any and all suits, actions or administrative proceedings, claims, demands, losses, damages, costs and expenses of whatsoever nature, including attorney's fees and expenses, which the Developer may suffer as a result of any infringement or alleged infringement of any patent, utility model, registered design, trademark, copyright or other intellectual property right registered or otherwise existing at the date of the Agreement by reason of the setting up of the Project by the Developer.

9.4 Monetary Limitation of Liability

9.4.1 A Party ("Indemnifying Party") shall be liable to indemnify the other Party ("Indemnified Party") under this Article 9 for any indemnity claims made in a Contract Year only up to an amount as per Article 4.1.6 of this Agreement.

9.5 Procedure to Claim Indemnity

9.5.1 Where the Indemnified Party is entitled to indemnification from the Indemnifying Party pursuant to Articles 9.1 or 9.2 the Indemnified Party shall promptly notify the Indemnifying Party of such claim, proceeding, action or suit referred to in Articles 9.1 or 9.2 in respect of which it is entitled to be indemnified. Such notice shall be given as soon as reasonably practicable after the Indemnified Party becomes aware of such claim, proceeding, action or suit. The Indemnifying Party shall be liable to settle the indemnification claim within thirty (30) days of receipt of the above notice.

Provided however that, if:

- (i) the Parties choose to contest, defend or litigate such claim, action, suit or proceedings in accordance with Article 9.5.3 below; and
- (ii) the claim amount is not required to be paid/deposited to such third party pending the resolution of the Dispute,
- (iii) the Indemnifying Party shall become liable to pay the claim amount to the Indemnified Party or to the third party, as the case may be, promptly following the resolution of the Dispute, if such Dispute is not settled in favour of the Indemnified Party.
- 9.5.2 The Indemnified Party may contest, defend and litigate a claim, action, suit or proceeding for which it is entitled to be indemnified under Articles 9.1 or 9.2 and the Indemnifying Party shall reimburse to the Indemnified Party all reasonable costs and expenses incurred by the Indemnified Party. However, such Indemnified Party shall not settle or compromise such claim, action, suit or proceedings without first getting the consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed.
- 9.5.3 An Indemnifying Party may, at its own expense, assume control of the defence of any proceedings brought against the Indemnified Party if it acknowledges its obligation to indemnify such Indemnified Party, gives such Indemnified Party prompt notice of its intention to assume control of the defence, and employs an independent legal counsel at its own cost that is reasonably satisfactory to the Indemnified Party.

9.6 Duty to Mitigate

The party entitled to the benefit of an indemnity under this Article 9 shall take all reasonable measures to mitigate any loss or damage which has occurred. If the party fails to take such measures, the other party's liabilities shall be correspondingly reduced.

10 Assignments and Charges

10.1 Assignments

Both the Parties shall at all times have the right to assign its rights, benefits and obligations under this Agreement and also convey, transfer and vest all the shares together with all beneficial interest in the shares to any entity or entities as Government of India / Nodal Agency may from time to time direct as a part of re-organization or restructuring of electricity industry provided that such successor entities expressly assume the obligations of the Party under this Agreement and are either an owner or operator of a substantial portion of the transmission system of India and are capable of performing the obligations of the concerned Party under this Agreement.

Subject to Article 10.2, this Agreement shall not be assigned by the Developer other than by mutual agreement between the parties.

10.2 Permitted Charges

- 10.2.1 Notwithstanding Article 10.1, the Developer may create any encumbrance over all or part of the Security Package or the other assets of the Transmission System / Project in favour of the Lenders or the Lender's representative as security for:
 - (a) Amounts payable under the Financing Agreements; and
 - (b) Any other amounts agreed by the parties.

Provided that:

(a) The Lenders or the Lender's Representative shall have entered into the Financing Agreements; and

(b) Any encumbrance granted by the Developer in accordance with this Article 10.2.1 shall contain provision pursuant to which the Lenders or the Lender's Representative agrees unconditionally with the Developer to release such encumbrances upon payment by the Developer to the Lenders of all amounts due under the Financing Agreements.

10.2.2 Article 10.2.1 does not apply to

- (a) Liens arising by operation of Law (or by an agreement evidencing the same)in the ordinary course of the Developer carrying out the TransmissionSystem/Project;
- (b) Pledges of goods, the related documents of title and / or other related documents arising or created in the ordinary course of the Developer carrying out the Transmission System/Project; and
- (c) Security arising out of retention of title provisions in relation to goods acquired in the ordinary course of the Developer carrying out the Transmission System/Project.

11 Governing Law and Disputes

11.1 Governing Law:

This Agreement shall be governed by and construed in accordance with the Laws of India. Any legal proceedings in respect of any matters, claims or disputes under this Agreement shall be under the jurisdiction of appropriate courts in Delhi.

11.2 Settlement of Disputes:

11.2.1 Either Party is entitled to raise any claim, dispute or difference of whatever nature arising under, out of or in connection with this Agreement, including its existence or validity or whether during the execution of the Project or after its completion and whether prior to or after the abandonment of the Project or breach of the Agreement by giving a written notice to the other Party, which shall contain:

- (i) a description of the Dispute;
- (ii) the grounds for such Dispute; and
- (iii) all written material in support of its claim.
- 11.2.2 The other Party shall, within thirty (30) days of issue of notice issued under Article 11.2.1, furnish:
 - (i) counter-claim and defences, if any, regarding the Dispute; and
 - (ii) all written material in support of its defences and counter-claim.
- 11.2.3 Within thirty (30) days of issue of notice by the Party pursuant to Article 11.2.1, if the other Party does not furnish any counter claim or defense under Article 11.2.2, or thirty (30) days from the date of furnishing counter claims or defence by the other Party, both the Parties to the Dispute shall meet to settle such Dispute amicably. If the Parties fail to resolve the Dispute amicably within thirty (30) days from the later of the dates mentioned in this Article 11.2.3, the Dispute shall be referred for dispute resolution in accordance with Article 11.3 and 11.4 of this Agreement

11.3 Adjudicator

The following provisions shall apply between the Parties in respect of any difference or dispute, which has not been resolved by mutual consultation as provided in Article 11.2.3;

- (a) The Parties shall appoint by mutual agreement an Adjudicator, in relation to technical matters, the Adjudicator shall be an Independent Engineer and in relation to financial matters, the Adjudicator shall be a financial expert of repute in their respective fields;
- (b) The Parties shall finalise the points of such disputes or differences or the terms of reference before referring such disputes or differences to the Adjudicator;
- (c) The Adjudicator will resolve or settle such disputes or differences as per reasonable standards within sixty (60) days of such disputes or differences

- being referred to him and convey his recommendations in writing to both the Parties;
- (d) The Adjudicator shall, subject to paragraph (e) below, determine which Party bears the costs (or a portion of the costs) of settling or determining such dispute or difference; and
- (e) If the Adjudicator settles or determines a dispute or differences in a manner entirely consistent with a Party's initial position on such dispute or difference, the Adjudicator shall determine that the other Party bears the costs of settling or determining the dispute or difference.

11.4 Arbitration

- 11.4.1 If either Party is dissatisfied with the Adjudicator's recommendation, or if the Adjudicator fails to give a recommendation within sixty (60) days of a dispute or difference being referred to him or if the Parties fail to appoint an Adjudicator or to finalise the terms of reference within sixty (60) days of the dispute or difference having first arisen, then either Party may within sixty (60) days of occurrence of either of the foregoing events give written notice to the other Party of its intention to refer the dispute or difference for arbitration, as herein provided.
- 11.4.2 Where neither Party has given notice of intention to commence arbitration of a dispute or difference within the period stated in Article 11.4.1, then the related recommendation of Adjudicator, if any, shall become final and binding.
- 11.4.3 Any dispute in respect of which a notice of intention to commence arbitration has been given, in accordance with Article 11.4.1 shall be finally settled by Arbitration, which Arbitration award shall be final and binding upon the Parties. Arbitration may be commenced prior to or after completion of the Transmission System/Project.
- 11.4.4 Any dispute submitted by a Party to Arbitration shall be heard by an Arbitration Tribunal composed of three arbitrators, in accordance with the provisions of the Arbitration and Conciliation Act, 1996 as amended or reenacted from time to time.

11.4.5 Notwithstanding anything provided in this Agreement to the contrary, any matter which is, by or under Electricity Act, 2003, directed to be determined by arbitration shall be determined in accordance with Section 158 of the Electricity Act, 2003, and the Rules and Regulations thereunder.

11.5 Parties to Perform Obligations:

Notwithstanding the existence of any Dispute and difference referred to the Commission or the Arbitration Tribunal as provided in Article 11.4.4 and save as the Commission or the Arbitration Tribunal may otherwise direct by a final or interim order, the Parties hereto shall continue to perform their respective obligations (which are not in dispute) under this Agreement.

12 Confidentiality

12.1 Confidential Information

Subject to Article 12.2, both Parties shall at all times during the continuance of this Agreement:

- (a) Use their reasonable endeavours to keep all information regarding the terms and conditions of this Agreement and any data or information acquired under or pursuant to this Agreement confidential and accordingly neither Party shall disclose the same to any other person; and
- (b) Not use any document or other information obtained by them respectively by virtue of this Agreement concerning the other's undertaking for any purpose other than performance of the Party's obligations and exercise of its rights under this Agreement.

Provided that the provisions of this Article shall not apply to information, which at the time of disclosure was in the public domain other than by breach of the foregoing obligations of confidentiality.

12.2 Disclosure of Confidential Information

- 12.2.1 Each of the Parties shall hold in confidence the agreements relating to the Transmission System/Project and all documents and other information which is of a confidential nature disclosed to it by or on behalf of the other Party relating to the Transmission System/Project and shall not, save as may be required by Law or appropriate regulatory or statutory authorities, or to any Indian Governmental Instrumentality, or to prospective Lenders to, or Investors in the Developer or to the professional advisers of the Parties or of those prospective Lenders or Investors, publish or otherwise disclose or use the same for its own purposes otherwise than as may be required to perform its obligations under this Agreement.
- 12.2.2 Either Party shall be entitled to disclose the terms and conditions of this Agreement and any data or information acquired by it under or pursuant to this Agreement without the prior written consent of the other Party if such disclosure is made in good faith;

13 Miscellaneous Provisions

13.1 Language

- 13.1.1 All agreements, all correspondences and communications as to be given and all other documentation to be prepared and supplied under the Agreement shall be written in English, and the Agreement shall be construed and interpreted in accordance with that language.
- 13.1.2 If any of the agreements, correspondence or communications are prepared in any language other than English, the English translation of such documents, correspondence or communications shall prevail in matters of interpretation.

13.2 Amendment

- 13.2.1 This agreement shall not be amended except by written agreement between the Parties.
- 13.2.2 This Agreement is intended solely for the benefit of the Parties and their respective successors and permitted assigns.

13.3 Entirety

This Agreement along with its sections, schedules and appendices is intended by the Parties as the final expression of their agreement and is intended also as a complete and exclusive statement of the terms of their agreement.

13.4 Successors and Assigns

This Agreement shall be binding upon, and inure to the benefit of the Parties and their respective successors and assigns.

13.5 Notices

13.5.1 Save as otherwise expressly provided in this Agreement, all notices or other communications which are required or permitted under this Agreement shall be in writing and sufficient if delivered personally or sent by registered or certified mail, fax, addressed as follows:

If to the SPV: [Insert Address with Contact Person]

If to the Developer: [Insert Address with Contact Person]

- 13.5.2 All Notices or communications given by fax shall be confirmed by depositing a copy of the same in the post office in an envelope properly addressed to the appropriate Party for delivery by registered or certified mail. All Notices shall be deemed delivered upon receipt.
- 13.5.3 Any Party may by notice of at least fifteen (15) days to the other Party change the address and/or addresses to which such notices and communications to it are to be delivered or mailed.

13.6 Compliance with Law

Despite anything contained in this Agreement if any provisions of this Agreement shall be in deviation of, inconsistent with or repugnant to the provisions contained in the Electricity Act, 2003 as amended or re-enacted from time to time, such

provisions shall be deemed to be amended to the extent required to bring it into compliance with the relevant provisions of the Electricity Act, 2003 as amended and re-enacted from time to time.

IN WITNESS WHEREOF, THE PARTIES HAVE CAUSED THIS AGREEMENT TO BE EXECUTED BY THEIR REPRESENTATIVE/DULY AUTHORISED OFFICERS AS OF THE DATE SET FORTH ABOVE.

WITNESS:	
Signature: Name: Designation: Address:	For and on behalf of [Insert Name of SPV] at[Place] on [Date]
Signature: Name: Designation: Address:	Signature: Name: Designation: Address:
Address.	For and on behalf of [Insert Name of Developer] at[Place] on [Date]
	Signature: Name: Designation: Address:

1	[mn	lementation	Agreement
J	шир	icilicilitation	1 igicoment

Draft for Discussion

SCHEDULE 1

Scheduled COD:

Expiry Date:

STANDARD

AUTHORIZATION AGREEMENT

FOR

DEVELOPMENT AND OPERATION OF INTER-STATE TRANSMISSION SYSTEM

FOR TRANSMISSION OF ELECTRICITY THROUGH TARIFF BASED COMPETITIVE BIDDING

BETWEEN THE
[Insert Name of Nodal Agency]

AND

DESIGNATED ISTS CUSTOMERS (DICs)

.....2017

AUTHORIZATION AGREEMENT

This Authorization Agreement (hereinafter referred to as "Agreement"	or "th	е	
Agreement" or "this Agreement") is made on the [Insert D)ay] c	þ	
, [Insert Month] of Two Thousand and[Insert Year]	by an	d	
between the persons whose names and registered offices are set	out i	n	
Schedule 1 including their successors and permitted assigns.			

WHEREAS

- A. Ministry of Power, Government of India has issued directions to the Designated ISTS Customers (hereinafter referred to as "DIC" individually and "DICs" collectively) for authorising Nodal Agency for acting as counterparty on their behalf for the purpose of discharging its duties in terms of the Bidding Guidelines and those mentioned herein.
- B. In accordance with the Standard Bidding Guidelines, all the existing DICs have to authorise the Nodal Agency to act on their behalf including executing Implementation Services Agreement (hereinafter referred to as the "ISA"). Further, any new DIC shall also execute a Supplementary Agreement as per Schedule 2 of this Agreement authorising the Nodal Agency to act on its behalf including executing ISA.

- C. The Bidding guidelines envisage that the Nodal Agency shall carry out its responsibilities and duties during the construction, development and operational phase of the project and shall monitor the progress of the project with due support from the Central Electricity Authority (hereinafter referred to as ("CEA").
- D. The Nodal Agency shall be entitled to recover its expenses from the DICs for performing the obligations and activities under the ISA and as per the Bidding Guidelines and provisions of this Agreement.
- E. Further, in accordance with the Bidding Guidelines and this Agreement, the Nodal Agency shall be completely indemnified and held harmless from and against any and all damages, losses, liabilities, obligations, claims of any kind, including any tax liability, interest, costs, fees or expenses (including, without limitation, reasonable attorney's fees and expenses) suffered, incurred or paid, directly or indirectly, as a result of, in connection with, or arising out of the Nodal Agency's obligations of acting as the nodal agency and in its responsibility arising out of this Agreement and as per the Bidding Guidelines.
- F. In case of inconsistency between the terms & conditions of this Agreement and the Central Electricity Regulatory Commission (Sharing of Inter-State Transmission Charges and Losses) Regulations, 2010 {hereinafter referred to as "POC Regulations"} the conditions stipulated in the POC Regulations by the Commission shall prevail.

Articles

1 Definitions and Interpretations

1.1 Definitions

- 1.1.1 The words / expressions used in this Agreement, unless as defined below or repugnant to the context, shall have the same meaning as assigned to them by the Electricity Act, 2003 and the rules or regulations framed there under including those issued / framed by the Commission (as defined hereunder), as amended or re-enacted from time to time or the General Clauses Act, failing which it shall bear its ordinary English meaning.
- 1.1.2 The words/expressions when used in this Agreement shall have the respective meanings as specified below:
 - "Act" or "Electricity Act" or "Electricity Act 2003" shall mean the Electricity Act, 2003 and any amendments made to the same or any succeeding enactment thereof;
 - "Bill" shall mean any bill raised by the Nodal Agency on the DICs to recover the expenses in fulfilling the obligations as per Article 5.1 of this Agreement and in accordance with the Billing and Collection Procedure as may be specified by Commission from time to time;
 - "Bulk Power Transmission Agreement" or "BPTA" shall mean the agreement between the ISTS Licensees and DICs of the ISTS under the pre-existing arrangements for ISTS development and operations;
 - "Business Day" shall mean a day other than Sunday or a statutory holiday, on which the banks remain open for business in the State in which the Nodal Agency's registered office is located and the concerned DICs are located;
 - "CEA" shall mean the Central Electricity Authority constituted under Section 70 of the Electricity Act;
 - "Commission" or "CERC" shall mean the Central Electricity Regulatory Commission referred to in sub-section (1) of Section 76 of the Electricity Act, 2003 or its successors and assigns;
 - "Competent Court of Law" shall mean the Supreme Court or any High Court, or any tribunal or any similar judicial or quasi-judicial body in

India that has jurisdiction to adjudicate upon issues relating to the Project;

- "Consultation Period" shall mean the period of sixty (60) days or such longer period as the Parties may agree, commencing from the date of issue of Nodal Agency's Preliminary Termination Notice, as provided in Article 7.2.1 of this Agreement, for consultation between the Parties to mitigate the consequence of the relevant event having regard to all the circumstances;
- "Construction Period" shall mean the period from (and including) the Effective Date of the Implementation & Service Agreement up to (but not including) the COD of the Element of the Project in relation to an Element and up to (but not including) the COD of the Project in relation to the Project;
- "CTU" or "Central Transmission Utility" shall mean the Power Grid Corporation of India Limited notified as the CTU by the Central Government under Section 38 of the Electricity Act, 2003, or any other entity that may be notified by the Central Government from time to time;
- "Day" shall mean a day starting at 0000 hours and ending at 2400 hours;
- "Designated ISTS Customers" or "DICs" shall have the meaning as ascribed in the Central Electricity Regulatory Commission (Sharing of Inter-State Transmission Charges and Losses) Regulation 2010 and as amended or modified from time to time;
- "Dispute" shall mean any dispute or difference of any kind between the Nodal Agency and the DICs, in connection with or arising out of this Agreement including any issue on the interpretation and scope of the terms of this Agreement as provided in Article 8.1;
- "Effective Date" for the purposes of this Agreement, shall have the same meaning as per Article 3.1.1 and 3.1.2 of this Agreement;
- "Event of Default" shall mean the events as defined in Article 7 of this Agreement;
- "Financial Year" shall mean a period of twelve months at midnight Indian Standard Time (IST) between 1st April & 31st March;
- "Force Majeure" and "Force Majeure Event" shall have the meaning assigned thereto in Article 6.3;

"GOI" shall mean Government of India;

"Implementation and Service Agreement" or "ISA" shall mean agreement executed by Nodal Agency on behalf of DICs with the TSP in accordance with Standard Bidding Guidelines for development of ISTS;

"Implementing Agency" shall mean the agency designated by the Commission to undertake the computation of allocation of transmission charges and losses at various nodes/zones for the application period along with other functions mandated under Sharing Regulations or as may be assigned by the Commission from time to time;

"Indian Governmental Instrumentality" shall mean Government of India, Government of any State in India or any ministry, department, board, authority, agency, corporation, commission under the direct or indirect control of Government of India or any State Government or both, any political sub-division of any of them including any court or Commission or tribunal or judicial or quasi-judicial body in India but excluding the CTU, TSP, Nodal Agency and the Designated ISTS Customers;

"Indian Electricity Grid Code" or "Grid Code" or "IEGC" shall mean the Grid Code specified by the Central Commission under Clause (h) of sub-section (1) of Section 79 of the Electricity Act;

"Inter-State Generating Station" or "ISGS" shall mean a Central/other generating station in which two or more states have shares and whose scheduling is to be coordinated by the RLDC;

"Inter-State Transmission System" or "ISTS" shall mean Inter-State Transmission System and shall include

- i) Any system for the conveyance of electricity by means of a main transmission line from the territory of one State to another State
- ii) The conveyance of energy across the territory of an intervening State as well as conveyance within the State which is incidental to such interstate transmission of energy
- iii) The transmission of electricity within the territory of State on a system built, owned, operated, maintained or controlled by CTU.

- "Inter-State Transmission System Licensee" or "ISTS Licensee" shall mean the person having been awarded a Transmission License by the CERC under Section 14 of the Act;
- "Law" or "Laws" in relation to this Agreement, shall mean all laws including electricity laws in force in India and any statute, ordinance, rule, regulation, notification, order or code, or any interpretation of any of them by an Indian Governmental Instrumentality having force of law and shall include all rules, regulations, decisions and orders of the Commission;
- "Month" shall mean a period of thirty (30) days from (and excluding) the date of the event;
- "Notification" shall mean any notification, issued in the Gazette of India;
- "Operating Period" for any Element of the Project shall mean the period from (and including) the COD of such Element of the Project, up to (and including) the Expiry Date and for the Project, shall mean the period from (and including) the COD of the Project, up to (and including) the Expiry Date;
- "Preliminary Termination Notice" shall mean Nodal Agency's Preliminary Termination Notice as defined in Article 7.2.1 of this Agreement;
- "Project" shall mean development, construction, commissioning and operations of ISTS element, system, or network as the case may be.
- "Prudent Utility Practices" shall mean the practices, methods and standards that are generally accepted internationally from time to time by electric transmission utilities for the purpose of ensuring the safe, efficient and economic design, construction, commissioning, operation, repair and maintenance of the Project and which practices, methods and standards shall be adjusted as necessary, to take account of:

- (i) operation, repair and maintenance guidelines given by the manufacturers to be incorporated in the Project,
- (ii) the requirements of Law, and
- (iii) the physical conditions at the Site;
- (iv) the safety of operating personnel and human beings;
- "Regional Load Despatch Centre (RLDC)" means the Centre established under sub-section (1) of Section 27 of the Act;
- "Sharing Regulations" shall mean the Central Electricity Regulatory Commission (Sharing of Inter-State Transmission Charges and Losses) Regulations, 2010 and as amended from time to time;
- "Standard Bidding Guidelines" or "Bidding Guidelines" shall mean the "Tariff Based Competitive Bidding Guidelines for Transmission Service" and "Guidelines for Encouraging Competition in Development of Transmission Projects" issued by Government of India, Ministry of Power dated 17.4.2006 under Section 63 of the Electricity Act and as amended from time to time
- "STU" or "State Transmission Utility" shall be the Board or the Government company, specified as such by the State Government under sub-section (1) of Section 39 of the Electricity Act 2003;
- "Supplementary Agreement" shall mean an agreement to be executed by Nodal Agency and New DICs as per Schedule 2 of this Agreement.
- "**Termination Notice**" shall mean a Nodal Agency's Termination Notice given by the Nodal Agency to the DIC pursuant to the provisions of Article 7.2.4 of this Agreement for the termination of this Agreement;
- "Term of Agreement" for the purposes of this Agreement shall have the meaning ascribed thereto in Article 3.1.3 of this Agreement;
- "Transmission Service Agreement" or "TSA" shall be the agreement executed between the relevant parties pursuant to Regulation 2 (u) of the Central Electricity Regulatory Commission (Sharing of Inter State Transmission Charges and Losses) Regulation 2010 and as amended from time to time;

"Transmission Services Provider" shall mean the person executing the ISA with Nodal Agency as per the Bidding guidelines for implementation of the Project.

"Unforeseen Event" shall mean an event, situation or similar circumstances not contemplated or referred to in this Agreement, which could not have been foreseen by a prudent and diligent person, and the impact of which shall destroy the sub-stratum and economic foundation of the Project.

1.2 Interpretation:

Save where the contrary is indicated, any reference in this Agreement to:

"Agreement" shall be construed as including a reference to its Schedules, Appendices and Annexures:

"encumbrance" shall be construed as a reference to a mortgage, charge, pledge, lien or other encumbrance securing any obligation of any person or any other type of preferential arrangement (including, without limitation, title transfer and retention arrangements) having a similar effect;

"indebtedness" shall be construed so as to include any obligation (whether incurred as principal or surety) for the payment or repayment of money, whether in present or future, actual or contingent;

"person" shall have the meaning as defined in Section 2 (49) of the Act:

"winding-up", "dissolution", "insolvency", or "reorganization" in the context of a company or corporation shall have the same meaning as defined in the Companies Act, 1956/ Companies Act, 2013 (as the case may be).

- 1.2.1 Words importing the singular shall include the plural and vice versa.
- 1.2.2 This Agreement itself or any other agreement or document shall be construed as a reference to this or to such other agreement or document as it may have been, or may from time to time be, amended, varied, novated, replaced or supplemented.
- 1.2.3 A Law shall be construed as a reference to such Law including its amendments or re-enactments from time to time.

- 1.2.4 A time of day shall, save as otherwise provided in any agreement or document be construed as a reference to Indian Standard Time.
- 1.2.5 Different parts of this Agreement are to be taken as mutually explanatory and supplementary to each other and if there is any inconsistency between or among the parts of this Agreement, they shall be interpreted in a harmonious manner so as to give effect to each part.
- 1.2.6 The tables of contents and any headings or sub-headings in this Agreement have been inserted for ease of reference only and shall not affect the interpretation of this Agreement.
- 1.2.7 The words "hereof" or "herein", if and when used in this Agreement shall mean a reference to this Agreement.

2 PARTIES TO THE AGREEMENT

2.1 Existing DICs

- 2.1.1 The DICs as mentioned in the Schedule 1 of this Agreement, enter into this Agreement and bind themselves to the terms of this Agreement as on the date of signing of this Agreement.
- 2.1.2 In the event of any conflict between the existing Bulk Power Transmission Agreements (BPTA) or Transmission Service Agreement (TSA) and this Agreement, the terms of this Agreement shall prevail.
- 2.1.3 An ISGS who is not a signatory to a BPTA or TSA too may enter into this Agreement and bind itself to the terms of this Agreement

2.2 New DICs

- 2.2.1 This Agreement shall be signed by all new DICs intending to connect to and/or use the Inter-State Transmission System (ISTS).
- 2.2.2 Any new DIC may become a party to this Agreement subject to and in accordance with the provisions of this Agreement by way of execution of Supplementary Agreement, as annexed at Schedule 2, by such DICs and the Nodal Agency.
- 2.2.3 Upon execution of the Supplementary Agreement by both Nodal Agency and the new DICs in accordance with Article 2.2.2 of this Agreement, the new DICs shall become a Party to this Agreement.

2.3 Authorisation

Each DIC hereby irrevocably and unconditionally authorizes Nodal Agency to execute on behalf of such Party any Supplementary Agreement duly executed by the new DIC, as the case may be, and to thereby admit the new DIC, as the case may be, as a Party to this Agreement. Agreements as such executed, shall be deemed to have been signed by all the existing Parties of this Agreement. A copy of such Supplementary Agreement shall be sent by the Nodal Agency to all the DICs.

3 EFFECTIVENESS AND TERM OF AGREEMENT

3.1 Effective Date

- 3.1.1 This Agreement shall be effective from the date of signing of this Agreement.
- 3.1.2 Any new DIC shall be Party to this Agreement from the date of signing of the Supplementary Agreement, as executed between the Nodal Agency and the new DIC, as the case may be.
- 3.1.3 The Agreement shall remain operative till the same is renewed/replaced/modified.

4 NODAL AGENCY'S AND DIC'S OBLIGATION

4.1 Nodal Agency's Obligation

The Nodal Agency shall be responsible to ensure development and operation of the Project by carrying out the following activities.

- 4.1.1 Execute ISA with TSP.
- 4.1.2 To collect contract performance guarantee (CPG) from selected bidder as per Article 3 of ISA and refund CPG after COD has been achieved in terms of Article 6.5.1 of the ISA.
- 4.1.3 To monitor timely completion of the conditions subsequent by the TSP in accordance with Article 3.1.3 of the ISA
- 4.1.4 Monitor progress of works, project milestones as per ISA and ensure such works are carried out as per the prevalent regulations.
- 4.1.5 To coordinate the availability of the Interconnection Facilities to enable the TSP to connect the Project;
- 4.1.6 To provide assistance to arbitrators as they may require for performance of their duties as per the ISA and make representations on behalf of DICs during dispute resolution and arbitration proceedings.
- 4.1.7 To levy liquidated damages on TSP for failure to achieve project milestones and COD as per timelines specified in ISA.
- 4.1.8 To monitor the Project after COD to ensure operational availability and ensure it is maintained and operated by TSP in accordance with Article 7.1 of the ISA.

- 4.1.9 To ensure that Insurance cover has been taken by TSP in accordance with Article 9.1 and Article 9.2 of the ISA.
- 4.1.10 To examine claims of Change in Law by TSP as per Article 12.3 of the ISA.
- 4.1.11 Any other obligations under ISA and Bidding Guidelines.

4.2 DIC's Obligation

4.2.1 To support Nodal Agency in discharging its duties and functions under this Agreement and ISA as and when required.

5 BILLING AND COLLECTION

5.1 Billing and Collection Mechanism

- 5.1.1 The Nodal Agency shall Bill to the DICs its expenses incurred towards fulfilment of its obligations under this Agreement and ISA as per the detailed "Billing and Collection Procedure" as approved by the Commission and such "Billing and Collection Procedure" shall be construed as part of this Agreement.
- 5.1.2 The Nodal Agency shall raise Bills and collect in accordance with the detailed "Billing and Collection Procedure".

5.2 Payment by DICs

The Payment to be done by DICs shall be governed as per the detailed "Billing and Collection Procedure" as approved by the Commission.

6 FORCE MAJEURE AND UNFORESEEN EVENT

The following terms shall have the meanings given hereunder.

- 6.1 An "Affected Party" means any of the DICs or the Nodal Agency whose performance has been adversely affected by an event of Force Majeure.
- 6.2 An "Unforeseen Event" shall mean an event, situation or similar circumstances not contemplated or referred to in this Agreement, which could not have been foreseen by a prudent and diligent person, and the impact of which shall destroy the sub-stratum and economic foundation of the Project.
- 6.3 A 'Force Majeure' means any event or circumstance or combination of events and circumstances including those stated below that wholly or partly prevent or unavoidably delay an Affected Party in the performance of its obligations under this Agreement, but only if and to the extent that such events or circumstances are not within the reasonable control, directly or indirectly, of the Affected Party and could not have been avoided if the Affected Party had taken reasonable care or complied with Prudent Utility Practices:

6.3.1 Natural Force Majeure Events:

Act of God, including, but not limited to drought, fire and explosion (to the extent originating from a source external to the Site), earthquake, volcanic eruption, landslide, flood, cyclone, typhoon, tornado, or exceptionally adverse weather conditions, which are in excess of the statistical measures for the last hundred (100) years.

6.3.2 Non-Natural Force Majeure Events:

- a) Direct Non-Natural Force Majeure Events
- (i) Nationalization or compulsory acquisition by any Indian Governmental Instrumentality of any material assets or rights of the Affected Party; or
- (ii) the unlawful, unreasonable or discriminatory revocation of, or refusal to renew, any Consents, Clearances and Permits required by the Affected Party to perform their obligations under the Agreement including the Standard Bidding Documents.
- b) Indirect Non Natural Force Majeure Events

- (i) act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo, revolution, riot, insurrection, terrorist or military action; or
- (ii) radio active contamination or ionising radiation originating from a source in India or resulting from any other Indirect Non Natural Force Majeure Event mentioned above, excluding circumstances where the source or cause of contamination or radiation is brought or has been brought into or near the Site by the Affected Party or those employed or engaged by the Affected Party; or
- (iii) industry-wide strikes and labour disturbances, having a nationwide impact in India.

6.4 Force Majeure Exclusions

- 6.4.1 Force Majeure shall not include (i) any event or circumstance which is within the reasonable control of the Parties and (ii) the following conditions, except to the extent that they are consequences of an event of Force Majeure:
- 6.4.2 Strikes or labour disturbance at the facilities of the Affected Party; Insufficiency of finances or funds or the Agreement becoming onerous to perform; and
- 6.4.3 Non-performance caused by, or connected with, the Affected Party's: negligent or intentional acts, errors or omissions; failure to comply with an Indian Law; or breach of, or default under this Agreement or any Project Documents.

6.5 Notification of Force Majeure Event

6.5.1 The Affected Party shall give notice to the other Party of any event of Force Majeure as soon as reasonably practicable, but not later than seven (7) days after the date on which such Party knew or should reasonably have known of the commencement of the event of Force Majeure. If an event of Force Majeure results in a breakdown of communications rendering it unreasonable to give notice within the applicable time limit specified herein, then the Party claiming Force Majeure shall give such

notice as soon as reasonably practicable after reinstatement of communications, but not later than one (1) day after such reinstatement.

- 6.5.2 Provided that, such notice shall be a pre-condition to the Affected Party's entitlement to claim relief under this Agreement. Such notice shall include full particulars of the event of Force Majeure, its effects on the Party claiming relief and the remedial measures proposed. The Affected Party shall give the other Party regular reports on the progress of those remedial measures and such other information as the other Party may reasonably request about the Force Majeure.
- 6.5.3 The Affected Party shall give notice to the other Party of (i) the cessation of the relevant event of Force Majeure; and (ii) the cessation of the effects of such event of Force Majeure on the performance of its rights or obligations under this Agreement, as soon as practicable after becoming aware of each of these cessations.

6.6 Duty to perform and duty to mitigate

To the extent not prevented by a Force Majeure Event, the Affected Party shall continue to perform its obligations as provided in this Agreement. The Affected Party shall use its reasonable efforts to mitigate the effect of any event of Force Majeure as soon as practicable.

6.7 Available Relief for a Force Majeure Event

Subject to this Article 6,

- 6.7.1 No Party shall be in breach of its obligations pursuant to this Agreement to the extent that the performance of its obligations was prevented, hindered or delayed due to a Force Majeure Event;
- 6.7.2 Each Party shall be entitled to claim relief for a Force Majeure Event affecting its performance in relation to its obligations under this Agreement.
- 6.7.3 For so long as the DIC is claiming relief due to any Force Majeure Event under this Agreement, the Nodal Agency may, from time to time on one

(1) day notice, inspect the DICs and the DIC shall provide the Nodal Agency's personnel with access to carry out such inspections, subject to the Nodal Agency's personnel complying with all reasonable safety precautions and standard.

6.8 Relief for Unforeseen Event

- 6.8.1 Upon occurrence of an Unforeseen Event, the Party affected by the Unforeseen Event shall have the right to refer the same to the Commission along with documentary evidence of the occurrence of the Unforeseen Event.
- 6.8.2 The Commission shall decide whether:
 - a. An Unforeseen Event has occurred;
 - b. The effects of such Unforeseen Event cannot be mitigated without a remedy or relief which is not contemplated in the Agreement; and
 - c. The Unforeseen Event or its effects have not been caused by any Party by any act or omission on its part.
- 6.8.3 If the Commission is satisfied that each of the conditions specified hereinabove is fulfilled, it shall issue an order to this effect and if necessary revise the Tariff.

7 EVENTS OF DEFAULT AND TERMINATION

7.1 Event of Default of a DIC

- 7.1.1 The occurrence and continuation of any of the following events shall constitute a DIC event of default, unless any such DIC event of default occurs as a result of Force Majeure Event.
- 7.1.2 A DIC fails to comply with the prevailing regulations including the Indian Electricity Grid Code or is in material breach of this Agreement and such material breach is not rectified by the said DIC within thirty (30) days of receipt of notice in this regard from the Nodal Agency;
- 7.1.3 Any of the Representation and Warranties made by the DIC in Article 10 of this Agreement is found to be untrue and/or inaccurate; or

7.1.4 If,

- a) A DIC becomes voluntarily or involuntarily the subject of any bankruptcy or insolvency or winding up proceedings and such proceedings remain uncontested for a period of thirty (30) days; or
- b) Any winding up or bankruptcy or insolvency order is passed against a DIC;
 or
- c) A DIC goes into liquidation or dissolution or a receiver or any similar officer is appointed over all or substantially all of its assets or official liquidator is appointed to manage its affairs, pursuant to Law, provided that it shall not constitute a DIC Event of Default where such dissolution or liquidation of such DIC is for the purpose of a merger, consolidation or reorganization and where the resulting entity has the financial standing to perform its obligations under this Agreement, similar to such DIC and expressly assumes all obligations of such DIC under this Agreement and is in a position to perform them.

7.2 Cessation of DIC being Party to this Agreement

7.2.1 Upon the occurrence and continuance of a DIC Event of Default under Article 7.1, the Nodal Agency may serve notice on the concerned DIC ("Nodal Agency's Preliminary Termination Notice"), with a copy to the CERC, which shall specify in reasonable detail, the circumstances giving rise to such Notice.

Draft For Discussions

- 7.2.2 Following the issue of such notice, as mentioned in Article 7.2.1, the Consultation Period shall apply and Nodal Agency and the concerned DIC shall discuss as to what steps shall be taken with a view to mitigate the consequences of the relevant Event of Default having regard to all the circumstances.
- 7.2.3 During the Consultation Period, the DIC shall, save as otherwise provided in this Agreement, continue to perform its obligations under this Agreement.
- 7.2.4 Following the expiry of the Consultation Period, unless the Nodal Agency and the concerned DIC shall have otherwise agreed to the contrary or the circumstances giving rise to such notice as mentioned in Article 7.2.1 shall have ceased to exist or shall have been remedied, the concerned DIC shall cease to be a Party to this Agreement and the Nodal Agency shall issue a written notice ("Termination Notice") of thirty (30) days to this effect with a copy to the Commission and Implementing Agency. The concerned DIC shall cease to be a Party to this Agreement on the date of expiry of the Termination Notice.

8 GOVERNING LAW AND DISPUTE RESOLUTION MECHANISM

8.1 Governing Law:

This Agreement shall be governed by and construed in accordance with the Laws of India. Any legal proceedings in respect of any matters, claims or disputes under this Agreement shall be under the jurisdiction of appropriate courts in Delhi.

9 ASSIGNMENT AND CHARGES

9.1 Assignments:

This Agreement shall be binding upon, and inure to the benefit of the Parties and their respective successors and permitted assigns. This Agreement shall not be assigned by any Party, except as provided in Article 9.2

9.2 Permitted Charges

Neither the DICs nor the Nodal Agency can relinquish or transfer its rights and obligations, without prior approval of the Commission or Appropriate Authority.

10 REPRESENTATION AND WARRANTIES

- 10.1 The DIC hereby represents and warrants to and agrees with the Nodal Agency as follows and acknowledges and confirms that the Nodal Agency is relying on such representations and warranties in connection with the transactions described in this Agreement:
- 10.2 It has all requisite powers and has been duly authorized to execute and consummate this Agreement;
- 10.3 This Agreement is enforceable against it, in accordance with its terms;
- 10.4 The consummation of the transactions contemplated by this Agreement on the part of the DICs will not violate any provision of nor constitute a default under, nor give rise to a power to cancel any charter, mortgage, deed of trust or lien, lease, agreement, license, permit, evidence of indebtedness, restriction, or other contract to which the DIC is a Party or to which the DIC is bound which violation, default or power has not been waived;
- 10.5 The DIC is not insolvent and no insolvency proceedings have been instituted, nor threatened or pending by or against the DIC;
- 10.6 There are no actions, suits, claims, proceedings or investigations pending or, to the best of the DIC's knowledge, threatened in writing against the

DIC at law, in equity, or otherwise, and whether civil or criminal in nature, before or by, any court, commission, arbitrator or governmental agency or authority, and there are no outstanding judgments, decrees or orders of any such courts, commission, arbitrator or governmental agencies or authorities, which materially adversely affect its ability to comply with its obligations under this Agreement.

10.7 The DIC makes all the representations and warranties above to be valid as on the Effective Date of this Agreement.

11 LANGUAGE

11.1 Language:

- 11.1.1 All agreements, correspondence and communications between the Parties relating to this Agreement and all other documentation to be prepared and supplied under the Agreement shall be written in English, and the Agreement shall be construed and interpreted in accordance with English language.
- 11.1.2 If any of the agreements, correspondence, communications or documents are prepared in any language other than English, the English translation of such agreements, correspondence, communications or documents shall prevail in matters of interpretation.

12 SEVERABILITY

The invalidity or enforceability, for any reason, of any part of this Agreement shall not prejudice or affect the validity or enforceability of the remainder of this Agreement, unless the part held invalid or unenforceable is fundamental to this Agreement.

13 NO CONSEQUENTIAL OR INDIRECT LOSSES

- 13.1 The liability of the DIC shall be limited to that explicitly provided in this Agreement.
- 13.2 Provided that, notwithstanding anything contained in this Agreement, under no event shall the Nodal Agency or the DIC claim from one another any indirect or consequential losses or damages.

14 INDEMNIFICATION BY THE DIC'S

14.1 Indemnification

The DICs agree to indemnify and hold the Nodal Agency harmless from and against any and all damages, losses, liabilities, obligations, claims of any kind including any tax liability, interest, cost, fee, or expenses (including, without limitation, reasonable attorneys' fees and expenses) (collectively, "Losses"), suffered, incurred or paid, directly, as a result of, in connection with or arising out of exercise it's obligations pursuant to and in accordance with this Agreement along with the bidding guidelines.

14.2 Indemnity to Survive Termination

The obligation of the DICs under Clause 13 shall survive the termination of this Agreement.

IN WITNESS WHEREOF THE PARTIES HAVE CAUSED THIS AGREEMENT TO BE EXECUTED BY THEIR DULY AUTHORISED REPRESENTATIVE AS OF THE DATE AND PLACE AS HEREIN AFTER APPEARING

WITNESS:	
Signature: Name: Designation: Address:	For and on behalf of [Insert Name of Nodal Agency] at[Place] on[Date]
Signature: Name: Designation: Address:	Signature: Name: Designation: Address:
	For and on behalf of [Insert Name of DIC] at [Place] on [Date]
	Signature: Name: Designation: Address:

Schedule 1

- 1. Parties to the Agreement [to be amended from time to time]
- 1.1 Nodal Agency:[Insert the name of Nodal Agency].
- 1.2 List of Designated ISTS Customers

S. No.	Name	Category (ISGS/STU/SEB/Bulk Consumers)	Address of Registered Office	Telephone/Mobile No./e-mail/FAX
		,		

Note: Any addition or deletion in this list after the signing of this Agreement shall be duly notified to the Parties to this Agreement

Schedule 2

SUPPLEMENTARY AGREEMENT

OOT I ELINENTANT AGREEMENT
BETWEEN
[Insert Name of Nodal Agency]
AND
[Insert name of the new Designated ISTS Customer]
THIS SUPPLEMENTARY AGREEMENT entered into on [Insert day] of
AND

WHEREAS:

part.

A. Nodal Agency has executed the Authorization Agreement with all the Designated ISTS Customers, as listed out in Schedule 1 of this Agreement.

thereof include its successors, and permitted assigns) as Party of the second

B. The new Designated ISTS Customer has been granted long term access or medium term open access to the ISTS by the Central Electricity Regulatory Commission (CERC).

NOW THEREFORE THIS AGREEMENT WITNESSETH as under:

The new Designated ISTS Customer and its long term access or medium term open access details are as described below:

ame of the new Designated STS Customer	Details of Long Term Access/ Medium Tem Open Access	Scheduled date/period of Long Term Access/ Medium Term Open Access

- 2) The new Designated ISTS Customer agrees to fulfill the terms and conditions laid down in the Authorisation Agreement in accordance with the terms of the same and the provisions of this Supplementary Agreement.
- 3) All terms and conditions of the Authorisation Agreement between the Nodal Agency and the Designated ISTS Customers (as listed out in Schedule 1 of the Authorisation Agreement) shall apply, *mutatis mutandis* without any change, to the new Designated ISTS Customer.

IN WITNESS WHEREOF THE PARTIES HAVE EXECUTED THESE PRESENTS THROUGH THEIR AUTHORISED REPRESENTATIVES

WIINESS:	
1. Signature:	For & on behalf of Nodal Agency
Name:	
Designation:	
2. Signature:	For and on behalf of new
Name:	Designated ISTS Customer)
Designation:	